Climate Services xxx (xxxx) xxx-xxx



Contents lists available at ScienceDirect

Climate Services



journal homepage: www.elsevier.com/locate/cliser

Spatially enabling the Global Framework for Climate Services: Reviewing geospatial solutions to efficiently share and integrate climate data & information

Gregory Giuliani^{a,b,*}, Stefano Nativi^d, Andre Obregon^e, Martin Beniston^a, Anthony Lehmann^{a,c}

^a University of Geneva, Institute for Environmental Sciences, Bd Carl-Vogt 66, CH-1205 Geneva, Switzerland

^b University of Geneva, Institute for Environmental Sciences, GRID-Geneva, Bd Carl-Vogt 66, CH-1211 Geneva, Switzerland

^c University of Geneva, Forel Institute, Bd Carl-Vogt 66, CH-1205 Geneva, Switzerland

^d National Research Council of Italy – Institute of Atmospheric Pollution Research, Via Madonna del Piano 10, Sesto Fiorentino (FI), Italy

^e GEO Secretariat, 7 bis avenue de la Paix, Case postale 2300, CH-1211 Geneva 2, Switzerland

ARTICLE INFO

Keywords: Climate services Essential Climate Variables Interoperability OGC standards GFCS, GEO/GEOSS

ABSTRACT

In November 2016, the Paris Agreement entered into force calling Parties to strengthen their cooperation for enhancing adaptation and narrowing the gap between climate science and policy. Moreover, climate change has been identified as a central challenge for sustainable development by the United Nations 2030 Agenda for Sustainable Development.

Data provide the basis for a reliable scientific understanding and knowledge as well as the foundation for services that are required to take informed decisions. In consequence, there is an increasing need for translating the massive amount of climate data and information that already exists into customized tools, products and services to monitor the range of climate change impacts and their evolution. It is crucial that these data and information should be made available not in the way that they are collected, but in the way that they are being used by the largest audience possible.

Considering that climate data is part of the broader Earth observation and geospatial data domain, the aim of this paper is to review the state-of-the-art geospatial technologies that can support the delivery of efficient and effective climate services, and enhancing the value chain of climate data in support of the objectives of the Global Framework for Climate Services. The major benefit of spatially-enabling climate services is that it brings interoperability along the entire climate data value chain. It facilitates storing, visualizing, accessing, processing/analyzing, and integrating climate data and information and enables users to create value-added products and services.

1. Introduction

According to the WMO Statement on the Status of the Global Climate in 2015 (World Meteorlogical Organization, 2016), the globally averaged temperature over land and ocean surfaces for 2015 was the highest among all years since record keeping began in 1880. This is also the largest margin by which the annual global temperature has been reached. During the 21st Conference of Parties (COP) of the United Nations Framework Convention on Climate Change (UNFCC) in Paris in 2015, countries have renewed their commitments to continue their efforts against global warming with the aim to limit the increase if possible to a maximum of 1.5 °C above preindustrial levels (UNFCCC, 2015). This requires a profound transformation moving towards resilient low carbon economies and implies substantial reductions of fossil fuel emissions of 80–95% by 2050; a complete phase-out by 2100; and significant adaptation efforts (Christoph et al., 2016).

To support this objective and continue to narrow the gap between climate science and policy, the Paris Agreement that entered into force in November 2016 calls Parties to strengthen their cooperation on enhancing action on adaptation. In particular, Articles 7(a) and (c) emphasize the need of improved sharing of information and "Strengthening scientific knowledge on climate, including research, systematic observation of the climate system and early warning systems, in a manner that informs climate services and supports decision-making" (UNFCCC, 2015). As a consequence, there is an increasing need for translating the massive amount of climate data and information that already exists into customized tools, products and services to monitor the range of climate change impacts and their evolution (Dolman et al., 2016).

* Corresponding author at: University of Geneva, Institute for Environmental Sciences, EnviroSPACE Lab., Bd Carl-Vogt 66, CH-1211 Geneva, Switzerland. *E-mail address*: gregory.giuliani@unige.ch (G. Giuliani).

http://dx.doi.org/10.1016/j.cliser.2017.08.003

Received 13 December 2016; Received in revised form 21 August 2017; Accepted 23 August 2017

2405-8807/ © 2017 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/BY-NC-ND/4.0/).

G. Giuliani et al.

The importance of climate change as a central challenge for sustainable development has been reinforced by the United Nations 2030 Agenda for Sustainable Development (United Nations, 2012, 2015). This agenda is a plan for action articulated around 17 Sustainable Development Goals (SDGs). Climate change has been identified as one of the greatest global challenges our society is facing today. It is considered as a cross-cutting challenge in various SDGs that can undermine the ability of countries to achieve sustainable development, putting billions of people at risk and in particular the most vulnerable communities in less developed countries. Specifically related to climate change, Goal 13 is calling for urgent action to combat climate change and its impacts, while enhancing resilience of our societies to natural hazards and climate change and developing a sustainable low-carbon economy.

To address the Paris Agreement and SDGs challenges, timely and reliable access to data and information on the environment, how it evolves is essential. Data provide the basis for a reliable scientific understanding and knowledge as well as the foundation for services that are required to take informed decisions (Trenberth et al., 2016).

To answer this need, the Global Framework for Climate Services (GFCS) has been established by the United Nations and spearheaded by the World Meteorological Organization (WMO) to support the development and application of science-based climate information and services for effective decision-making (see Table 1 for a glossary of terms used in this paper). For the GFCS, climate services involve the production, translation, transfer, and use of climate data and information to support climate-informed decision-making, policy development and planning (World Meteorlogical Organization, 2011). The ultimate objective is to ensure that the best available climate science is effectively used and communicated to various sectors (e.g., agriculture, water, health) that may benefit from climate knowledge (Lucio et al., 2016). This requires accessing reliable national and international repositories of data such as temperature, precipitation, wind, soil moisture, or ocean

Table 1

A glossary of terms used in this paper.

| conditions as well as assessments, projections, scenarios or vulner- |
|---|
| ability and risk analyses. Moreover, following users' needs, these data |
| may be combined with other types of data like socioeconomic variables, |
| health trends, and agricultural production or energy production (Swart |
| et al., 2017). |

Currently, the demand for climate services is lower than what is required to deliver the expected benefits so that the potential market is largely unrealized (Lourenço et al., 2016; Street, 2015). One of the main challenges that climate services have to face is to reduce the gap between climate science and decision-makers (Vaughan et al., 2016). Indeed, on the one hand, climate scientists are generally interested in an improved understanding of the processes that regulate the climate, while on the other hand, decision-makers need simple and easy to obtain knowledge for informed decision making. This has led to a disconnection between real and perceived needs of climate knowledge. Consequently, to close this gap, it is essential to have proper engagement and interactions between providers and users of climate information together with an effective data access mechanism to answer various user needs (Buontempo et al., 2014).

Achieving the objective of a sustainable development requires the integration of different data sets on physical, chemical, biological, and socio-economic systems coming from various sources (Lehmann et al., 2017). Collectively, these diverse data constitute a set of environmental attributes describing a specific location; they can therefore be considered to be part of geospatial data. Environmental data are valuable when combined with other data sets, e.g., social and economic, allowing one to monitor and assess the status of global, regional or local environments, to discover relationships between them, or to model future changes. To make sense of the huge amount of environmental data that exists and that is currently generated on a daily basis, it is essential to agree upon common standards to facilitate their sharing and integration. It is in this context that the concept of Spatial Data Infrastructure (SDI) has emerged. This term, first introduced by the U.S.

| Term | Context |
|---|---|
| Climate data (series) | A time series of measurements of sufficient length, consistency, and continuity to determine climate variability and change (Committee on Climate Data Records from NOAA Operational Satellites, 2004). These data series can be generated by in situ measurements (e.g. ground-based sensor measurements), remote sensing (e.g., satellite observations) and models (e.g., predictions and projections) |
| Climate services | Climate information that assists decision making by individuals and organizations ^a |
| Essential Climate Variables (ECV) | An ECV is a physical, chemical or biological variable or a group of linked variables that critically contributes to the characterization of Earth's climate ^b and support policy action |
| Data value chain | Information flow that describes a series of steps needed to generate value and useful insights from data (European Commission DG Connect, 2014). These steps includes: enhanced data discovery (e.g., capture, storage, organization), integration (e.g., visualization, access), and exploitation (e.g., transformation, analysis, tailored products and services) |
| Interoperability | The extent to which systems and devices can exchange data, and interpret that shared data. For two systems to be interoperable, they must be able to exchange data and subsequently present that data such that it can be understood by a user ^c |
| Standard | A document that provides rules or guidelines to achieve order in a given context ^d . In the domain of Information and Communication Technologies (ICT), standards address especially the needs for interconnection and interoperability. Standards are frequently referenced by regulators and legislators for protecting user and business interests, and in support of government policies. |
| Web service | A collection of operations offered by a provider to users, using the World Wide Web for communicating |
| Sustainable Development Goals (SDG) | A set of 17 goals to end poverty, protect the planet, and ensure prosperity for all as part of the United Nations Sustainable Development $Agenda^e$ |
| Group on Earth Observation (GEO) | The Group on Earth Observations (GEO) was established in 2005 as an intergovernmental mechanism for coordinating all existing and future Earth observations systems and implementing a "Global Earth Observation System of Systems" (GEOSS). It was launched in response to calls from the WSSD, the G8 and three ministerial Earth Observation Summits to improve existing Earth observation systems |
| Global Earth Observation System of Systems (GEOSS) | GEOSS is a set of coordinated, independent Earth observation, information and processing systems that interact and provide access to diverse information for a broad range of users in both public and private sectors. GEOSS links these systems to strengthen the monitoring of the state of the Earth. It facilitates the sharing of environmental data and information collected from the large array of observing systems contributed by countries and organizations within GEO |

^a http://www.wmo.int/gfcs/what_are_climate_weather_services.

^b https://public.wmo.int/en/programmes/global-climate-observing-system/essential-climate-variables.

^c http://www.himss.org/library/interoperability-standards/what-is-interoperability.

^d http://www.etsi.org/standards/what-are-standards.

^e http://www.un.org/sustainabledevelopment/sustainable-development-goals/.

National Research Council (National Research Council & Mapping Science Committee, 1993) denotes a framework to facilitate and coordinate provision and exchange of geospatial data encompassing different data sources, systems, networks, standards and organizations policies. The main goal is to deliver geospatial data and information from many diverse sources to the widest possible group of potential users. In 2005, the Group on Earth Observation (GEO¹) was launched to build a global platform for sharing environmental data across several so-called Societal Benefit Areas (SBA) such as climate and weather. In 2007, the European Union (EU) adopted a directive (i.e. the INSPIRE Directive) to create a EU Spatial Data Infrastructure (European Commission, 2007). Several environmental disciplines (e.g., biodiversity, oceanography, hydrology) that have similar concerns regarding data accessibility, availability, compatibility, and integration have already embraced SDI concepts (Lehmann et al., 2014a,b). From a technological perspective, in the Web era, the fundamental condition to facilitate data exchange and integration is to agree on the use of open and standard protocols (i.e., operation interfaces and data encoding schemas) that enable systems interoperability to exchange and use information (Geraci, 1991; Open Geospatial Consortium, 2004).

Comparatively to other environmental sciences, climate-science is shifting more slowly to standard-based approaches (Woolf et al., 2005). It is still common to visualize maps and graphs as simple images and sharing data statically using FTP sites or download pages with various data formats. This lack of interoperability impedes efficient and effective integration of climate data and information with other data, products or applications. Consequently, it is difficult to answer the different needs and requirements of various climate data users and the value chain of climate data is not yet fully realized.

To tackle this data integration issue, it is essential to extend the definition of climate services with some architectural principles to ensure the provision of interoperable, flexible and efficient services and facilitate the discovery, access and use of climate-science data to a wide range of potential users. Considering that climate data is part of the broader Earth observation and geospatial data (United Nations, 2015), the aim of this paper is to review the state-of-the-art geospatial technologies that can support the delivery of efficient and effective climate services, enabling the value chain of climate data in support of the GFCS objectives (Fig. 1).

The remainder of the paper is structured as follows. Section 2 presents the GFCS and its objectives. Section 3 discusses the current status for delivering climate services by presenting various international, regional and national initiatives as well as some research projects. Section 4 presents the methodology for selecting the literature reviewed of geospatial standards and emerging technological concepts that can support the delivery of efficient and effective climate services. Section 5 expands the notion of service with a technological perspective and review challenges and possible solutions to better store and share climate data (5.1), improve data discovery (5.2), deal with data quality and uncertainty (5.3), cope with data visualization and download (5.4), improve data processing (5.5), facilitate integration (5.6), develop tailored products (5.7), and develop capacities (5.8). Based on this review, the Section 6 presents selected examples of increased use of interoperability in climate science. The Section 7 discusses benefits, limitations and perspectives of spatially-enabling the GFCS. Finally, the paper closes with conclusions and suggestions for more interoperable climate services.

2. The Global Framework for Climate Services (GFCS)

In 2009, at the Third World Climate Conference, the idea of a framework engaging climate scientists along with providers and users of climate data and information emerged. Experts were recognizing that

current capabilities and mechanisms were not meeting present and future needs for providing effective access to science-based climate information (World Meteorlogical Organization, 2011). Consequently, thirteen heads of state or government, 81 ministers and 2500 scientists agreed to establish the GFCS supervised by the WMO to strengthen the production, availability, delivery of science-based climate information and services and to coordinate the development, implementation and application of climate products to support decision-making (Huges, 2011). The vision of the $GFCS^2$ is to enable society to better manage the risks and opportunities arising from climate variability and change, especially for those who are most vulnerable to such risks. This must be done through the development and incorporation of science-based climate information and prediction into planning, policy and practice. The greatest value of the GFCS should occur incrementally through the delivery of a multitude of climate services at various scales (e.g., from local to global).

Several drawbacks in climate data delivery and integration have been identified and these findings have served as foundations for the GFCS (World Meteorlogical Organization, 2011). It has been recognized that:

- (1) Climate services do not efficiently exploit scientific climate knowledge, information and data;
- (2) Climate services do not meet present and future user needs and in particular in developing or least developed countries that are the most vulnerable;
- (3) Providers of climate services do not interact sufficiently with users;
- (4) Existing capacities for climate observations provide a good basis for strengthening climate services but commitment to sustain highquality observations across the entire climate system is inadequate;
- (5) Enhancements in observations networks in developing countries are required;
- (6) Restrictions about the sharing and access to data and information are a major barrier to progress and wide use of climate knowledge;
- (7) Use of climate knowledge that can inform decision-making is inadequate and is not following the rapid advancement of the understanding of climate system; and
- (8) Capacities of users is often insufficient to adequately use climate data and information.

The High-Level Taskforce on the GFCS defines climate services as a range of activities that is aiming at generating and providing information in a way that assists decision-making by individuals and organizations based on past, present and future of climate and on its impacts on natural and human systems (Lucio et al., 2016).

The central principle guiding the implementation of the GFCS is that it should serve the widest audience possible and in particular climatevulnerable developing countries. The objectives of the GFCS are (1) to improve climate services for all countries, (2) build capacities of providers and users, (3) enable governments to have a central role as primary sources of climate services, (4) promote a free and open exchange of climate data and information while respecting existing data policies, (5) facilitate the timely access to relevant scientific information to help society to cope with current climate variability and limit economic and social damages caused by climate-related disasters.

The priority areas³ for action are agriculture and food security, water management, disaster risk reduction, energy, and health. These five areas are recognized as key sectors where the GFCS can play a major role for climate change adaptation and mitigation and sustainable development. They are aligned with the needs and objectives of initiatives like the Sustainable Development Goals, the Hyogo Framework for Action, the Water-Food-Energy Nexus (Hoff, 2011), or the

¹ http://www.earthobservations.org.

² http://www.gfcs-climate.org/vision.

³ http://www.gfcs-climate.org/priority-areas.

Climate Services xxx (xxxx) xxx-xxx





UNFCCC (Hewitt et al., 2012).

To strengthen the provision and use of climate predictions, products, information, data from local to global scales, the GFCS^{4,5} relies on five main components (Fig. 2) (Hewitt et al., 2012):

The following height principles⁶ are being used to guide the implementation of the GFCS:

- High priority for the needs of climate-vulnerable developing countries;
- (2) Primary focus is the better access and use of climate information by users;
- (3) The Framework addresses needs at three spatial scales: global, regional and national;
- (4) Climate services must be operational and continuously updated;
- (5) Climate information is primarily an international public good and governments will have a central role in the Framework;
- (6) The Framework encourages global, free and open exchange of climate-relevant data;
- (7) The Framework facilitates and strengthens and does not duplicate efforts; and
- (8) The Framework is built through partnerships.

The GFCS is aiming at supporting and fostering collaboration between global, regional and national initiatives and stakeholders (Hewitt et al., 2012). At the global level, the objective is to define large-scale activities necessary for delivering and developing climate services. At the regional scale, the focus is on strengthening cooperation and engaging with multilateral efforts to address regional needs (e.g., infrastructure development, data and information exchange, capacity building, research and training). Finally, at the national scale, the GFCS aims at ensuring effective coordination mechanisms by each national government to involve all relevant stakeholders to indicate their requirements for climate services.

To ensure that climate services answer users' needs an appropriate mechanism needs to be established that require technical capacities and active communication and exchange between data and information producers, translators, and community of users. Central to this task is the former Climate Information and Prediction Services (CLIPS). So far, CLIPS was the tool used by WMO to provide access to climate information. It has been recently closed and assimilated in the GFCS within the Climate Services Information System (CSIS) (Srinivasan et al., 2015). CLIPS has various elements that can be useful for the CSIS. It provides data management facilities as well as delivery mechanism for accessing climate monitoring and assessments, climate predictions and projections. It also provides a dedicated user interface for climate adaptation and risk management. However, this Information System



Fig. 2. The components of the GFCS and their relations (Adapted from World Meteorological Organization, 2011). The User Interface Platform is aiming at enabling users, researchers, and providers of climate services to interact to maximize the significance of climate services and lead to the development of applications that answers user needs. The Climate Services Information System should provide an effective mechanism to access and distribute climate data and information needed by users in accordance with rules, procedures, and restrictions agreed by data providers. The Observations and Monitoring component should ensure the generation climate observations necessary to meet the needs of climate services users and providers. The Research, Modeling, and Predictions component will assess and promote the development of climate prediction tools, products and the promotion of climate services in research agendas. Finally, the Capacity Building component is an overarching component aiming at supporting systematic capacity development of institutional, infrastructure and human resources.

does not implement any interoperability arrangements.

Crucial for the success of the GFCS is building partnerships and obtaining the active engagement of major organizations by involving and receiving strong commitments from governments and agencies in the governance and implementation processes. The major challenge for GFCS is to gain recognition from governments that climate services have substantial value and deserve support (Hewitt et al., 2012). In particular, due to the global scale of climate change and its relationship with sustainable development, GFCS requires an international coordination. In its initial phase, the GFCS concentrated on developing and delivering services in the five priority areas where improved access to science-based climate knowledge can have immediate impacts in improving human safety and well-being (Hewitt et al., 2012). In term of governance, the main body is the Intergovernmental Board on Climate Services (IBCS), supported by a Management Committee for carrying out the decisions and requests of the Board; a Partner Advisory Committee (PAC) for stakeholder engagement; and a technical committee. To start developing capabilities at national and regional levels and engaging with user communities, the GFCS is supporting several

⁴ http://www.wmo.int/pages/themes/climate/climate_services.php.

⁵ http://www.gfcs-climate.org/components-of-gfcs.

⁶ http://www.gfcs-climate.org/principles.

projects (http://www.wmo.int/gfcs/projects-map) with a particular attention on six countries (Bhutan, Burkina Faso, Dominica, Moldova, Papua New Guinea and Tanzania). This will help to highlight the benefits of collaboration and gather lessons learnt for replication of good practices in other countries. The ultimate objective is that by 2021, the GFCS should be fully operational and climate services should be widely used especially in climate-sensitive sectors.

3. Current situation in delivering climate services

3.1. Current situation at various scales

Currently, governments and agencies in every region of the world are already providing or developing a wide range of climate services for various sectors such as agriculture, water, health, disaster risk reduction, energy, transport and infrastructure, ecosystems, and urban issues (World Meteorlogical Organization, 2012). This wealth of services represents a solid foundation on which GFCS can rely to advance improvements in the provision of services tailored to users' needs. For example, the European Space Agency (ESA) Climate Change Initiative (CCI)⁷ or the Committee on Earth Observation Satellites (CEOS) Working Group Climate (WGClimate)⁸ are coordinating efforts to make full use of Earth Observations space assets to exploit long-term records on Essential Climate Variables (ECV) (Hollmann et al., 2013). In particular, the CEOS WGClimate has written a report on satellite observations for climate monitoring and the need for an international architecture that ensures timely delivery of observations for an efficient and effective analysis of the Earth's climate system (Dowell et al., 2013).

At the global scale, the Group on Earth Observations (GEO) Forest Carbon Tracking initiative9 (that is part of the broader Global Forest Observation Initiative (GFOI)) is integrating in-situ observations and remote sensing data to estimate forest cover and carbon content. The main objective is to provide a set of monitoring, reporting and verifications tools to support climate mitigation policies. Another initiative also led by GEO is the Global Agricultural Monitoring (GEOGLAM¹⁰) aiming at providing reliable, accurate, timely and sustained crop monitoring information and yield forecasts. Monitoring crop conditions within countries at risk of food security, the GEOGLAM Early Warning Crop Monitor¹¹ can contribute to track SDG Goal 2 to "End hunger, achieve food security and improved nutrition, and promote sustainable agriculture" and in particular target 2c to adopt measures to ensure the proper functioning of food commodity markets and their derivatives, and facilitate timely access to market information, including food reserves, to help limit extreme food price volatility.

At the regional scale, the European funded the PRUDENCE¹² and ENSEMBLES¹³ projects aimed to provide high resolution, quality controlled regional climate ensembles of predictions and link them to a wide range of applications (e.g., agriculture, health, water, food security). The European Climate Adaptation Platform (CLIMATE-ADAPT)¹⁴ is a joint effort between the European Commission (EC) and the European Environment Agency (EEA) to support Europe in adapting to climate change by helping users to access and share data and information on expected impacts; vulnerability of regions and activities; adaptation strategies, options, actions and tools. Recently launched, the Copernicus Climate Change Service (C3S)¹⁵ will be the major

contribution from the European Union to the GFCS in particular for the *Observations and Monitoring* component.

At the national scale, various countries have already embraced the concept of climate services to support National Adaption Plans (NAP) (i.e., the means for identifying medium to long-term adaptation needs and developing and implementing strategies to address these needs) like the United States Climate.gov¹⁶, the National Centre for Climate Services in Switzerland (NCCS)¹⁷, the German Climate Portal¹⁸, the Climate Service Center Germany (GERICS)¹⁹ or the Southern African Development Community Climate Services Centre (SADC CSC)²⁰. This indicates that at the three levels targeted by the GFCS several on-going initiatives are aiming to provide climate services.

3.2. Networks of end users, providers and researchers

All these efforts at different scales are complemented by interdisciplinary networks of climate information users, providers and researchers who are interested in climate services and are actively involved in the community. Such efforts are represented by the Climate Services Partnership²¹, JPI-Climate²², or CORDEX²³ (Giorgi et al., 2009). In addition, several research projects are complementing these institutional efforts for advancing the vision of climate services like IS-ENES2²⁴ for Earth System Modeling (Valcke et al., 2015), the IMPAC-T2C²⁵ on quantifying projected impacts under 2 °C warming (Preuschmann et al., n.d.), MARCO²⁶ for giving insights into the climate services market in Europe, ERA4CS²⁷ for boosting the development of efficient Climate Services (Kotova et al., 2017), or ANYWHERE²⁸ (EnhANcing emergencY management and response to extreme WeatHER and climate Events) to respond more rapidly than today to extreme climate and weather events. For additional examples of services and delivery structures readers can refer to Vaughan and Dessai (2014) and Medri et al. (2012).

3.3. Main climate data standards used

These examples demonstrate that the volume of climate data worldwide is expanding rapidly, thereby generating challenges in term of archiving, sharing, discovering, accessing and integrating it (Overpeck et al., 2011). These data usually documents past, present and future conditions of the climate system and are generated by models (e.g., predictions, projections), remote sensing (e.g., satellite observations), and in situ measurements. Most climate data are encoded using the netCDF (Network Common Data Format) model and format, defined by Unidata, following the CF (Climate and Forecast) conventions to formalize the necessary semantics as metadata (Domenico and Nativi, 2012). Alternatively, ASCII file formats are well adopted and the File Transfer Protocol (FTP) is widely used to transfer data. Climate data are commonly visualized as tables, graphs or maps. Overpeck et al. (2011) stated that the two major pressing issues for climate science concerning data are (1) ensuring the ever-growing volumes of data are easily and freely accessible to facilitate new scientific research, and (2) guarantee that these data and the information and knowledge generated from them are useful and understandable by a large interdisciplinary

⁷ http://cci.esa.int/data.

⁸ http://ceos.org/ourwork/workinggroups/climate/current-activities/.

⁹ http://www.gfoi.org/rd/forest-carbon-tracking/.

¹⁰ http://www.earthobservations.org/geoglam.php.

¹¹ http://www.geoglam-crop-monitor.org.

¹² http://prudence.dmi.dk/main.html.

¹³ http://ensembles-eu.metoffice.com.

¹⁴ http://climate-adapt.eea.europa.eu/about.

¹⁵ http://climate.copernicus.eu.

¹⁶ https://www.climate.gov.

¹⁷ http://www.meteoswiss.admin.ch/home/research-and-cooperation/nccs.html.

¹⁸ http://www.deutschesklimaportal.de/EN/Home/home_node.html.

¹⁹ http://www.climate-service-center.de.

²⁰ http://www.sadc.int/sadc-secretariat/services-centres/climate-services-centre/.

²¹ http://www.climate-services.org/.

²² http://www.jpi-climate.eu/home.

²³ http://www.cordex.org.

²⁴ https://is.enes.org.

²⁵ http://impact2c.eu

²⁶ http://marco-h2020.eu.

²⁷ http://www.jpi-climate.eu/ERA4CS.

²⁸ http://anywhere-h2020.eu.

G. Giuliani et al.

audience.

To monitor and understand the causes of climate change, identify potential impacts, evaluate options for adaptation and mitigation, and characterize extreme events (e.g., floods, droughts) global high-quality comprehensive and coordinated observations are necessary. Without such a baseline, it will be difficult to meet the various climate data user's requirements and developed tailored products for policy maker or stakeholders (Sessa and Latham, 2008). Moreover, these data are not only essential for climate science but they are also required for other environmental and sustainability disciplines (e.g., Ecosystem Services, Sustainable Development Goals, Nexus).

3.4. The Global Climate Observing System and Essential Climate Variables

The Global Climate Observing System (GCOS) is a joint undertaking of WMO, UNEP, the Intergovernmental Oceanographic Commission (IOC) of the United Nations Educational Scientific and Cultural Organization (UNESCO) and the International Council for Science (ICSU). GCOS has been established in 1992 to support IPCC with the objective of ensuring that observational data required for addressing climate-related issues are available to all potential users (Ostensen et al., 2008). GCOS has identified 50 Essential Climate Variables (ECVs²⁹) that are required to support the work of the UNFCCC and the IPCC. These ECVs are technically and economically feasible for systematic observation (Bojinski et al., 2014). They can be measured using satellite remote sensing or terrestrial, atmospheric, and oceanographic sampling (Ostensen et al., 2008). These ECVs are essential data for delivering climate services and they are already widely adopted within science and policy domains (Bojinski et al., 2014). To facilitate and provide an effective access mechanism to ECVs, GCOS has initiated the Global Observing Systems Information Center (GOSIC) (Diamond and Lief, 2009). The GOSIC portal³⁰ is a clearinghouse (i.e., does not hold data) acting as a centralized entry-point to access data and information for GCOS. It enables users to search and access data in a harmonized and integrated way across multiple data repositories. Various matrices have been developed to provide rapid overviews and access to relevant data (Diamond and Lief, 2009) for both terrestrial and oceanic systems. Most interestingly, GOSIC has also designed an ECV Data Access Matrix³¹.

Even if this represents an important step towards supporting data access for climate services, this ECV Data Access Matrix presents several drawbacks. It helps data users to identify repositories, however it is probably not enough to provide a service that can meet different expectations from various user communities. Indeed, it does not provide any search functionality, it is not possible to visualize the datasets, users have to deal with various data formats (e.g., NetCDF, ASCII file, CSV, GeoTIFF) and various data access methodologies (e.g., FTP, Web Accessible Folders, RESTful services, download pages), data are often not in the same spatial coordinate projection and temporal resolution and therefore should be transformed to be ready to be ingested in a model (e.g. data processing) or to be integrated with other data to better assess environmental conditions in a given region. In other words, this requires a lot of work before analyzing data and transform into useful information. Several studies (European Commission, 2015; Overpeck et al., 2011; Street, 2015) have identified data and information sharing and their related technical challenges as essential to strengthen the provider-user interface (Vaughan et al., 2016). Their aim is to provide reliable, usable, fit-for-purpose services that will help integrating climate information with multiple data sources within a multi-disciplinary framework.

³⁰ https://www.ncdc.noaa.gov/gosic.

3.5. Climate services and the geospatial community

Considering the fact that climate data are describing a location/ region of the Earth through a set of attributes, they can be thought of as being part of geospatial data. In the geospatial community, a framework enabling connection between data providers and users is traditionally known as SDI. This promotes the use of a common (web-based) services bus that enables systems interoperability. This set of common services facilitates discovery, access and use of data for the whole infrastructure. Service-based interoperability is achieved through the recognition and implementation of well-defined interface protocols, which are specified by international (or sometimes Community) standards. Consequently, many of the international standards (i.e. OGC, ISO, and at a certain extent W3C) as well as the current research in the field of system-of-systems interoperability (e.g. software ecosystems) can significantly help GFCS to strengthen the provider-user interface and enhance the quality and relevance of climate services.

In the next section, we will review significant geospatial standards and emerging technological concepts that could benefit to spatiallyenabled the GFCS, expand the notion of service with a technological perspective, enhance the quality and relevance of climate services, and help deliver efficient services that can be used in a multi-disciplinary framework.

4. Methodology

The review presented in this paper follows the general methodology proposed for systematic reviews (Collaboration for Environmental Evidence, 2013; Pullin and Stewart, 2006). This type of review is particularly useful when a subject is the focus of considerable research in recent years and where a comprehensive view can be useful for orienting future research and methods (Plummer et al., 2012). This is the case for climate services and in particular of accessing climate data in various disciplines. The objective of this study is to get a picture of the use of geospatial standards in relation with climate services and capture what are the benefits, limitations and perspectives in using geospatial technologies to support the delivery of efficient and effective climate services, and enhancing the value chain of climate data in support of the GFCS objectives.

The approach taken for searching relevant peer-reviewed and nonpeer-reviewed literature consisted of a set of keywords used to query different repositories such as scientific libraries (e.g., Science Direct, Web of Knowledge, Google Scholar), personal databases of the researchers and their research groups, and the Internet (e.g., Google searches). These searches were not limited by geographical extent or scale. The following list of keywords were used individually for each query: 'climate services', 'ogc', 'interoperability', 'essential climate variables', 'geoss', 'climate data' producing a comprehensive list of articles. To refine results three additional criteria were used: articles should address climate services as the main or secondary subject; the keywords should be at least in the title, keywords or abstract; and articles should be written in English. Finally, following the recommendations for Internet searches the first 50 records were examined while the subsequent 50 were looked at for relevance (Collaboration for Environmental Evidence, 2013).

The combined results of these various searches account for more than 250 references over the last two decades. About 120 papers were excluded because they were beyond the scope of this study. The remaining articles were filtered manually to avoid duplications and screened to ensure that they are relevant to the climate data value chain. The final list of papers (about 70) were then categorized according to identified challenges that prevent interoperability along the data value chain.

²⁹ https://www.wmo.int/pages/prog/gcos/index.php?name = EssentialClimateVariables.

³¹ https://www.ncdc.noaa.gov/gosic/gcos-essential-climate-variable-ecv-data-accessmatrix.

5. Spatially-enabling the GFCS through standards and data value chains

From an architectural point of view, a fundamental requirement for delivering efficient climate services is to ensure that datasets are compatible among data producers by adopting standard specifications supported by different organizations. It is only through consistent and harmonized datasets that a multi-organizations community can expect effective data, which is required to achieve user objectives by responding to their needs (Sessa and Latham, 2008).

Recognizing that climate data is an essential resource for environmental research and sustainability assessment, it is necessary to coordinate all actions in a coherent workflow (i.e. data value chain), from data acquisition to knowledge generation for decision making, and to serve a wide range of stakeholders by supporting their technologies (Miller and Mork, 2013).

Such a data value chain will help creating and building value through enhanced data discovery (e.g., acquisition, storage, management), integration (e.g., access, download, visualization), and exploitation (e.g., transformation, analysis, tailored products and services). To ensure that measurements, observations, model outputs and predictions are meaningful and sustainable, standard schemes of data content can help in documenting, searching, accessing, interpreting, and integrating climate data and ultimately strengthening the inter-operability interface between providers and users (Ostensen et al., 2008).

The Open Geospatial Consortium (OGC³²) and the International Organization for Standardization (ISO)/Technical Committee (TC) 211 Geographic Information/Geomatics³³ are the leading international organizations developing a suite of open standards for modeling and implementing geospatial information interoperability. These standards can pave the way to meaningful data interoperability in climate science and help delivering and leveraging the power of climate services building discovery, access and use services for climate datasets and information (Woolf et al., 2005; Woolf et al., 2006).

5.1. Struggling with data formats, storage and sharing

The first element to consider in the data value chain is related to data acquisition, storage and management. Climate datasets can be typically considered as "Big Data". Big data can be characterized according to their features represented in a four-dimensional space: Volume, Velocity, Variety and Veracity (Gijzen, 2013; Lee and Kang, 2015; UN Global Pulse, 2012). In terms of volume, Climate Data Centers usually handle Petabytes of data. These data arise from an almost continuous stream of diverse sources such as in-situ sensor measurements (e.g., temperature, precipitation, wind); remote sensing observations (e.g., satellite, aircraft, Unmanned Aerial Vehicle (UAV)), and climate models (e.g., predictions, projections).

As for variety, climate datasets are commonly encoded using different content models and formats, such as ASCII files, CSV (Comma Separated Values) files, and netCDF binary files for multi-dimensional and array-oriented scientific datasets. NetCDF, generally extended with the CF conventions, is particularly well suited, as it supports also the time dimension which is very important in climate science (Domenico and Nativi, 2012).

Finally, the veracity (e.g., data and model quality and uncertainty) of the huge amount of shared data is often an important issue to efficiently (re-)use climate data in the policy arena (Bastin et al., 2013). In particular, uncertainty is a major concern in ensemble forecasting, a commonly used method that involves multiple forecasts created with an individual forecast model by using different physical parameterizations

³² http://www.opengeospatial.org.

or varying initial conditions to generate a representative sample of the possible future states of weather or climate (Palmer, 2000).

Several solutions exist to facilitate the storage, management, publishing and sharing of Big Climate Data. For example, the Raster Data Manager (Rasdaman) is an array analytics engine built on top of the PostgreSQL Database Management System that helps storing, analyzing and publishing multidimensional arrays using various open standards (Baumann et al., 1999; Baumann et al., 2016). Alternatively, THREDDS – Thematic Real-time Environmental Distributed Data Services (Domenico et al., 2002) and GeoServer³⁴ can easily publish netCDF (and other data formats) and expose these datasets using open standards (Nativi et al., 2006).

5.2. Documenting data to strength data discovery

In the data value chain pattern, the primary action that users undertake is to search for data that are suitable for their purpose. Therefore, enabling users to find and evaluate datasets before accessing them is an essential task. Effective and efficient data discovery relies on the existence of catalogs containing quality (i.e., full and updated) information describing datasets to be shared (Giuliani et al., 2016a) (Santoro et al., 2012). This information is referred to as discovery metadata. The importance of having discovery metadata catalogs is emphasized by major data sharing initiatives such as GEOSS - Global Earth Observation Systems of Systems (Nativi and Bigagli, 2009), IN-SPIRE (European Commission, 2007) and NSDI (National Research Council, 1993). It was recently reinforced by Open Data policies (Wessels et al., 2014). All these initiatives highlight the need to use data description specifications (i.e. metadata standards) to document datasets for searching, evaluating, and using (i.e., discovery, evaluate, and use metadata types). Interoperable catalogs implement open and standard metadata schemas that can be exchanged and used by different data systems for diverse scopes.

Several open standard schemas can be used to describe a given geospatial dataset. The most commonly used are those developed by ISO TC211, such as the abstract data and service description models ISO 19115 (*Geographic information – Metadata*)³⁵ and ISO 19119 (*Geographic information – Metadata*)³⁵. Several profiles are based on these, like the Federal Geographic Data Committee (FGDC)³⁸ specifications and the INSPIRE metadata ones (European Commission, 2008). Alternatively, the Dublin Core Metadata Initiative³⁹ and the related W3 C DCAT⁴⁰ specifications are more general standards (referring to any online resource) that are also widely used.

Once data are properly documented, the documentation records (i.e. metadata records) should then be stored in a catalog software component, which provides standardized search and discovery operations. For interoperability sake, these catalogs are requested to expose at least a standard interface for exposing its geospatial records on the Web. The OGC Catalog Service for the Web (CSW) is a well-implemented standard that enables users to discover, browse, and query metadata describing geospatial data, services and other resource types (Open Geospatial Consortium, 2007a). Alternatively, OpenSearch APIs⁴¹ and its OGC profiles for geospatial information⁴² can be used. These different metadata models and catalog interface standards offer a

³³ http://www.isotc211.org.

³⁴ http://geoserver.org.

³⁵ http://www.iso.org/iso/catalogue_detail?csnumber=26020.

³⁶ http://www.iso.org/iso/catalogue_detail.htm?csnumber = 39890.

³⁷ http://www.iso.org/iso/catalogue_detail.htm?csnumber=32557.

³⁸ https://www.fgdc.gov/metadata/geospatial-metadata-standards.

³⁹ http://dublincore.org.

⁴⁰ https://www.w3.org/TR/vocab-dcat/.

⁴¹ http://www.opensearch.org/Home.

⁴² http://www.opengeospatial.org/standards/opensearchgeo.

coherent framework for data documentation and discovery.

Unidata THREDDS Data Server (TDS)⁴³ has several capabilities to expose netCDF metadata; one of them provides an ISO 19115 metadata representation of the datasets' structure and description (Chengfang et al., 2009). GeoServer offers similar capabilities through a CSW interface. Additionally, interesting work has been undertaken in the context of the EU FP7 project CHARMe⁴⁴ (Characterization of metadata to enable high-quality climate applications and services) that has developed the concept of commentary metadata. The objective is to link climate datasets with publications, citations, assessments, user feedback to help users learn from previous experiences and select the best datasets that can suit their needs as well as providing a direct traceability between conclusions and data that supported them (Blower et al., 2013). Finally, discovery enhancement by metadata augmentation (e.g. semantic annotation and inferences, user tagging and feedbacks, etc.) is a vibrant research and innovation area that is promising to significantly advance geospatial data discoverability.

5.3. Dealing with data quality and uncertainties

Facilitating data discovery and access is necessary but not sufficient. Data quality and uncertainty is essential to enhance communication, accountability, and support efficient decision-making processes (European Commission, 2015; Moges et al., 2016; Street, 2015). Climatology is dependent on heterogeneous data sources, therefore inherited errors (e.g., accuracy) and uncertainties due to the various methods (e.g., downscaling methods necessary to process the data at the geographic and temporal scales needed for impact studies) used to create datasets and run models are important topics to consider (Kave et al., 2012; Otto et al., 2015). Having means to ensure that quality and uncertainties are understood correctly can help informing adaptation and mitigation policies. Therefore, managing data and model uncertainty and quality in web-based frameworks is an important issue (Bastin et al., 2013). It is particularly true when we consider that policy and decision-makers are increasingly relying on scientific datasets and models to explore different scenarios to develop and/or take better informed decisions (Buytaert et al., 2012). Providing incomplete information can negatively influence decision-making and lead to the development of policies and strategies that are not responding the challenges we are facing (Otto et al., 2015). Consequently, effective instruments to quantify and efficiently communicate quality and uncertainties are essential.

As to quality, it is important to distinguish between a dataset producer's view (e.g., quality check and assurance of shared data) and user's view (e.g., information dealing with dataset fit-for-purpose, usability, and feedbacks from other users). For data quality, an important aspect to take into account is datasets maturity and applicability (i.e., which dataset is most useful for a specific application). A maturity model and matrix for climate data records have been proposed to ensure that basic measurements (e.g., raw data) are consistently transformed in quality-controlled, homogenized, and meaningful products (Bates and Privette, 2012). This model has been further developed by the EU FP7 project COordinating earth observation data validation for RE-analysis for CLIMAte Services (CORE-CLIMAX⁴⁵). The matrix gives information about how mature a dataset is in terms of metadata, quality, software, usage, documentation, uncertainty, and accessibility (EUMETSAT, 2014).

To tackle the issue of uncertainty in the Earth Observation domain, the EU recently funded several programs. UncertWeb (http://www. uncertweb.org) and GeoViQua (http://www.geoviqua.org) projects have introduced instruments (i.e., frameworks, models and tools) to

propose an interoperable representation of data uncertainties and quality. In particular, the Uncertainty Markup Language⁴⁶ (UncertML) and the Quality ML⁴⁷ are conceptual models and XML encoding designed to quantify, describe and exchange data quality and uncertainties (Diaz et al., 2012; Pebesma et al., 2011; Zabala et al., 2013). The OGC "NetCDF Uncertainty Conventions" (NetCDF-U) makes use of UncertML to formalize datasets uncertainty in a netCDF file (Bigagli and Nativi, 2013). In the framework of the EarthCube programme, the USA NSF has funded the "Advancing netCDF-CF for the Geoscience Community" project⁴⁸ which has been discussing on advanced CF conventions to document quality and uncertainty characterizing netCDF-CF datasets. GEOSS recently introduced a set of Data Management Principles⁴⁹ as a guideline for data providers to advance the quality of shared datasets. Among others, the EU funded projects Gap Analysis for Integrated Atmospheric ECV CLImate Monitoring (GAIA-CLIM⁵⁰), Fidelity and uncertainty in climate data records from Earth Observations (FIDUCEO⁵¹), and Quality Assurance for Essential Climate Variables (QA4ECV⁵²) (Scanlon et al., 2015) are tackling the aspect of uncertainty of EO-based climate data.

Finally, concerning models and projections, various methods are used to visualize uncertainty data. However, most of them are not supported in a standardized and consistent approach (Blower et al., 2015; Kristin et al., 2009). In particular, regarding model ensembles, UncertML and NetCDF-U appear interesting solutions as exemplified by the work done by ECMWF in the frame of the UncertWeb project⁵³ to represent probability distributions.

5.4. Coping with data visualization and download

Data visualization is an important tool for both scientists and nonscientists. It allows understanding and handling large volume of environmental data and for communicating scientific results within, between and outside scientific communities (Blower et al., 2009; Harold et al., 2016). Various techniques exist to visualize climate and environmental data (Kaye et al., 2012; Nocke et al., 2008; Sun et al., 2012). Typical limitations identified by these authors are lack of interactivity, lack of transparency, and lack of interoperability. Moreover, the heterogeneity of climate data (e.g., spatial, temporal, multi-variate, gridded, station measurements, remote measurements) requires different visualization techniques (e.g., 2D-maps, 3D-globes, time-series graphs) (Nocke et al., 2008).

To facilitate the visualization and access of climate data to different users' audience, current standards can greatly help in building standard-based, interactive, interoperable, web-based visualization systems of four-dimensional climate data (Sun et al., 2012). The OGC Web Map Service (WMS) is a mature standard for serving graphical representation (images) of geospatial data (Open Geospatial Consortium, 2006a). This standard supports multi-dimensional representation of complex data (e.g., time, elevation). A data-publishing server like GeoServer can easily handle the publication of netCDF data and expose it as WMS service⁵⁴. This enables any client that implements this standard to interactively visualize raster (e.g., satellite images), vector (e.g., geospatial 2 D data such as a point, line or polygon), and multi-dimensional data (e.g., netCDF). Other solutions like THREDDS that has the

 $^{^{\}mathbf{43}}\,\mathbf{http://www.unidata.ucar.edu/software/thredds/current/tds/.}$

⁴⁴ http://charme.org.uk.

⁴⁵ http://www.coreclimax.eu.

⁴⁶ http://www.uncertml.org.

⁴⁷ http://qualityml.geoviqua.org.

⁴⁸ https://www.earthcube.org/group/advancing-netcdf-cf.

⁴⁹ https://www.earthobservations.org/documents/dswg/201504_data_management_ principles_long_final.pdf.

⁵⁰ http://www.gaia-clim.eu.

⁵¹ http://www.fiduceo.eu.

⁵² http://www.qa4ecv.eu.

⁵³ http://www.uncertweb.org/uploads/deliverables/

e790f9107ac313083b15cbf91f90a7aa0c8fa3ac.pdf.

⁵⁴ http://geoserver.geo-solutions.it/multidim/en/index.html.

GODIVA2 viewer is also able to publish and handle WMS (Blower et al., 2009). The proposed WMS-Q standard (http://www.geoviqua.org/WMS-Q.htm) adds quality and uncertainty information (Blower et al., 2015) and visualize them with clients like Greenland⁵⁵, ncWMS-Q or MiraMon⁵⁶.

Concerning station measurements that are usually providing access to point-measurement time-series, the OGC Sensor Observation Service (SOS) is of high interest (Laney et al., 2015). It also enables users to integrate this type of climate data in an interoperable web-based framework.

To access data, the Web Feature Service (WFS) and the Web Coverage Service (WCS) enable users to download 2D georeferenced data (Open Geospatial Consortium, 2006b; 2010b).

OPeNDAP⁵⁷ is a popular standard technology (i.e. data model, software client and server components) in the oceanography community, to access climate and forecast datasets. It is based on the DAP2 protocol⁵⁸, which provides a discipline-neutral means of requesting and providing data across the Web-in particular, DAP2 provides a constraint notation for requesting parts of a multidimensional dataset.

Finally, to visualize statistics, the OGC Table Joining Service (TJS) is a standard that defines how to join tabular data (i.e., statistics) with a spatial reference (Open Geospatial Consortium, 2010a). This standard can help linking socio-economic and environmental data together (Grothe and Brentjens, 2013).

5.5. Improving data processing

Computing and IT infrastructures have been identified has a critical element to build the GFCS to facilitate data flow, transformation, analysis and processing of the exploding volume of climate data (Street, 2015). Different types of climate data sources (e.g., sensors, satellites, models) are required to develop and apply advanced data processing algorithms to describe the temporal evolution of climate. Moreover, the increasing volume of climate data generated at increasingly finer spatial and temporal resolution has made it necessary to use computing power that exceed the current capacities of single computers (Overpeck et al., 2011).

To tackle the issue of processing algorithms sharing and computing power, the OGC Web Processing Service (WPS) (Open Geospatial Consortium, 2007b) and the High-Performance and Distributed Computing paradigm (Buyya et al., 2009) can help.

WPS is a standard specification providing rules on how to invoke inputs and outputs as standardized web-based processing service. It defines how a client can request the execution of a process; how a provider can publish a processing algorithm as a service; and how the output of a process is handled. This enables providers and users to share algorithms in an interoperable way. However, to turn climate data into understandable information, efficient processing solutions are required. Distributed high performance computing infrastructures such a Super Computers, Clusters, Grids, and Clouds appear as promising solutions (Bosin et al., 2011; Shujia et al., 2010). Efforts are underway to unleash the power of computing infrastructures through interoperable webbased processing services (Giuliani et al., 2012; Mazzetti et al., 2016; Rodila and Gorgan, 2012; Rodila et al., 2015). The objective of such integration is to enable efficient analysis in a transparent and interoperable way, hiding the complexity of the infrastructure while letting users concentrating on analyses. WPS can play an important role also in the future distributed processing and analytics frameworks by supporting relevant patterns such as the Model-as-a-Service and the Online Analytical Processing (OLAP) interfaces -see for example, the GEO

Model Web (Nativi et al., 2013c) and the Data Cube infrastructure interfaces (Gray et al., 1995).

5.6. Facilitating integration and multi-disciplinarity

Another challenge to enhance the quality and relevance of climate data rests in the integration and framing of climate data with other sources to support decision-making (European Commission, 2015; Street, 2015). Decisions related to the climate require the integration of multiple sets of complex information about physical, chemical, biological, and socio-economical systems often provided by different organizations. Moreover, climate data can be useful for other scientific (and non-scientific) communities of practice. This raises the necessity of a collaborative, multi-disciplinary, and multi-stakeholder effort to provide an integrated access to climate resources. To support this effort it is necessary to find a consensus and ensure a given level of harmonization, while recognizing the diversity of stakeholders (both datasets providers and users) with their different aspirations and mandates (Strobl et al., 2011). Probably the best attempt to build a multi-disciplinary framework based on existing systems is represented by the System of Systems engineering approach exemplified by GEOSS (Béjar et al., 2009; Nativi et al., 2013b). GEOSS recognizes the heterogeneity of systems that reflect the diversity of stakeholders and decided not to impose a common and limited set of specifications to the different providers and users. Instead, GEOSS aims to act as a broker between data providers and users (Nativi et al., 2012, 2013a, 2015). Indeed, in the Earth science community there are different interoperability interfaces, tools and standards, thus making it very difficult to build a multidisciplinary framework for non-experts. To address these issues, GEOSS has developed a brokering framework enabling the binding of heterogeneous resources published by different data providers. The brokering framework implements all the mediation and adaptation tasks necessary to map the different standards, used by the heterogeneous communities, onto a common harmonized interface, which is completely transparent for both the data providers and users.

This facilitates linkages between communities allowing them to search, discover, and access heterogeneous resources, while allowing data users and providers to continue using their tools and publishing their resources with their usual standards. Such a framework can facilitate commitment, endorsement and acceptance on interoperability and creating the basis for a multi-disciplinary and integrative research. This fit very well with the needs of climate policy that requires a suitable science-policy interface to ensure that the best climate science is adequately transferred to decision-makers or other users. Finally, such an integrative framework can also facilitate interconnection/coupling of models as demonstrated by GEO Model Web initiative (Nativi et al., 2013a,b,c) that helped to link Climate Change and Biodiversity infrastructures (Nativi et al., 2009) or by the Open Modeling Interface (OpenMI) (Castronova et al., 2013; Goodall et al., 2013; Laniak et al., 2013).

5.7. Develop tailored applications and products

Once data and metadata are published and accessible through interoperable web services, it greatly facilitates the development of tailored applications that can answer users' needs. For example, webbased frameworks such as D3JS (https://d3js.org) enables manipulating documents based on data combining powerful visualization techniques and a data-driven approach. Depending on users' requirements, it allows generating different dynamic and interactive views (e.g., in the form of graphs or maps) of a same dataset. With such frameworks, applications are usable on both desktop and mobile devices. They also allow the development of dashboards than can be defined as a readable, intuitive, and interactive user interface showing a graphical presentation as a graph and/or a map of the current status and trends of a variable or indicator to enable immediate and informed decisions to

⁵⁵ https://wiki.52north.org/bin/view/Geostatistics/Greenland.

⁵⁶ http://www.creaf.uab.es/miramon/Index_usa.htm.

⁵⁷ https://www.opendap.org/about.

⁵⁸ https://www.opendap.org/pdf/dap_2_data_model.pdf.

be made. Dashboards can be a powerful tool to monitor ECVs or other climate-related datasets and to track their evolution through time. These kinds of products can certainly help to raise the value of climate data and services as well as reducing the gap between scientists and decision-makers.

5.8. Developing capacities and enhancing data policies

One of the objectives of the GFCS is to facilitate the development of the market for climate services that deliver social and economic benefits at various scales (Street, 2015). To support this objective, the GFCS encourage global, free and open exchange of climate-relevant data. Currently, there are several ongoing open data policies for facilitating the access and use of public data like the Digital Agenda for Europe. Such policies can support a more effective and immediate translation of data into information by building interoperable user-driven services (Cuca, 2016). It is essential to realize the value that open data can have on economy (e.g., broad benefits and growth), on society (e.g., enhanced social welfare), on research and innovation (e.g., new types of research, reproducibility, accountability), on education, and on governance (e.g., transparency) by encouraging the use and uptake of data (CODATA, 2015; Ryan, 2016). It can increase the use of - as well as bring important returns on - investment, potentially of several orders of magnitude higher than the initial investment. Probably, one of the best examples of the success of a broad and open access policy is the NASA Landsat program (National Geospatial Advisory Committee - Landsat Advisory Group, 2014). In 2008, the US government decided to completely open up the Landsat archive. Prior to this data, Landsat were sold at a price ranging between \$500-\$5000. At the peak of data sales in 2001, a mean of 52 scenes per day were sold corresponding to a revenue of \$4.5-5 million. Immediately after the open data policy, data access increased to a mean of 5700 scenes per day for a revenue estimated in 2011 to around \$2 billion (Ryan, 2016). Such policies can help to develop value-added products and services, increase transparency, and offer promising business opportunities and help national economies. From a scientific perspective, if such policies are adopted globally, it will completely transform our understanding how the Earth System works facilitating large scale access and integration of terrestrial, oceanic, and atmospheric data provided by government agencies.

However, to fully realize the potential of open data policies together with interoperable access to data, a key enabler is to develop capacities at human, institutional, and technical levels. It helps to raise awareness and create commitments on the benefits of data sharing and publication with interoperable solutions. Integrated material such as the "Bringing GEOSS services into practice"⁵⁹ (Giuliani et al., 2016b) can help to lower the entry barriers for both data providers and users, facilitates the development of technical skills, and empowers people.

6. Selected examples of increased use of interoperability in climate science

Several projects and initiatives are embracing interoperability in climate science and thus contributing to build spatially-enabled climate services. This contributes to a gradual shift towards more open, inclusive and interoperable systems, increasing the climate-data value chain and closing the gap between data providers and users. A good example is the C-READ platform⁶⁰ of the Caribbean Community Climate Change Center. It leverages several open standards components to provide interoperable discovery, access, processing functionalities for the monitoring of climate and environmental changes in the Caribbean region, allowing different partner countries and organizations to combine their data together with historical data to provide value-added

information products to inform decision and policy makers. Härtwig and Müller (2013) demonstrated how to use interoperable access to data together with processing services to provide climate classification functionality. It helps to give an intuitive overview of the distribution of different climate conditions around the globe. Sun et al. (2012) used Google Earth to develop a web-based visualization for climate research. For these authors, the use of an interoperable solution increased significantly the awareness, usability and visibility of scientific results. Sweden and UK have also implemented several open standards to increase the reuse of scientific environmental and climate data (Klein et al., 2013; Stephens et al., 2012). They highlight the importance of scientific data sharing as a prerequisite for new scientific findings, knowledge, and services, supporting decision-making to the rapid environmental and socio-economic challenges. Woolf et al. (2006) have demonstrated how interoperability standards can support the publication of legacy data sources, thereby highlighting the benefits for enabling simple use of data with deprecated data formats. Bernard and Ostländer (2008) have demonstrated how beneficial standards can be to assess vulnerability to climate change. Other interesting examples that can be mentioned are: the European Centre for Medium-Range Weather Forecasts (ECMWF⁶¹) that uses OGC WCS and WPS to offer interoperable services to access over 50 Petabytes of operational and research data (Wagemann et al., 2017). The Copernicus Atmosphere Monitoring Service⁶² enables users to discover high volume of data related to atmosphere composition. The United Nations Environment Programme (UNEP) has used the brokering technology to developed interactive graphs that allow monitoring the evolution of a set of ECVs in realtime⁶³. Another nice example of a platform that make extensive use of interoperability arrangement to facilitate data discovery, access, process, and visualization of Europe's climate data and information, is the EU project Climate Information Portal (CLIPC⁶⁴). Finally the Climate Inspector⁶⁵ enables users to investigate climate change around the globe across space and time, looking at climate trends, variability and uncertainty and access maps and data (Inman, 2011; Wilhelmi et al., 2016).

All these examples emphasize that interoperability provides climate services that are more efficient, enhances data integration, as well as the development of tailored products and services. These systems enable developing synergies and improving communication between scientists, stakeholders and decision makers, greatly advancing climate data sharing and scientific research collaboration.

7. Discussion and perspectives

During the COP21, countries were invited to adopt the Open Data Charter⁶⁶ to open climate data for enhancing a collective action exemplified by "Data for Climate Action – a global open innovation initiative for climate resilience"⁶⁷. Similarly, the WMO Resolution 60 on data policies for climate data in support of the GFCS that was endorsed in 2015 is also calling for open sharing of climate data (World Meteorlogical Organization, 2015). This charter is based on the following principles: (1) open by default; (2) timely and comprehensive; (3) accessible and usable; (4) comparable and interoperable; (5) for improved governance and citizen engagement; (6) for inclusive development and innovation. Several contributions have emphasized the need for open sharing of quality-assured data and services underlying published scientific knowledge to support climate change research and policymaking (Kaye et al., 2012; Nocke et al., 2008; Otto et al., 2015; Overpeck et al., 2011;

⁶¹ http://www.ecmwf.int.

⁶² http://atmosphere.copernicus.eu.

⁶³ http://ede.grid.unep.ch/extras/graphs_list.php.

⁶⁴ http://www.clipc.eu.

⁶⁵ https://gisclimatechange.ucar.edu/inspector.

⁶⁶ http://opendatacharter.net.

⁶⁷ http://www.unglobalpulse.org/data-for-climate-action.

⁵⁹ http://www.geossintopractice.org.

⁶⁰ http://c-read.net.

Sessa and Latham, 2008; Street, 2015). Particularly, scientists developing climate services should work in close connection with decision makers to deliver services corresponding to their needs. From a technical perspective, interoperability is a key enabler to facilitate the access to climate data and information in an efficient way.

7.1. Benefits

The major benefit of spatially-enabling climate services is that it brings interoperability along the entire data value chain. It facilitates storing, visualizing, accessing, processing/analyzing, and integrating climate data and information and enables users to add create valueadded products and services. NetCDF-CF, likely the most common data format in climate science, is already an OGC standards (Domenico and Nativi, 2012) and most of data publication tools (e.g., GeoServer, TDS) supports netCDF data. Therefore, it is simple to publish these data through interoperable web services.

Within Climate Science community netCDF is recognized as a de facto standard for array-oriented scientific data. However, a data model that is tailored to the needs of one discipline may not be appropriate for sharing data to other disciplines (Nativi and Domenico, 2009). Consequently, netCDF file based sharing might not be the best solution, while spatial databases can provide more flexibility. Indeed, time series and especially real-time data are difficult to store in existing SDIs (Härtwig and Müller, 2013). While for data with no or low update frequency (e.g., land cover) the best way to store is in spatial databases like PostGIS, time-series measurements (e.g., precipitation) are mainly stored in netCDF files. From a user point of view, it would be much easier to have a custom spatio-temporal subset of the climate data extracted from a spatial database and to be downloaded in different formats. To tackle this issue and provide more flexibility, the OGC SOS standard is particularly suited. It allows storing time-series measurements in a spatial database: facilitate integration with other data types through a standardized interface; and enables data extraction (e.g., by geographical and/or temporal extent) in various formats. Such approach has been exemplified in flood protection (Cannata et al., 2015), ecological observations (Munoz et al., 2014), water quality monitoring (Horsburgh et al., 2010), air quality monitoring (Stasch et al., 2012). All these examples have shown an increased interoperability, usage, and facility to integrate time-series measurements with other data. Experiences to use the SOS standard for providing gridded climate data to agricultural applications (Chinnachodteeranun and Honda, 2016) and for climate modeling (Kutschera et al., 2011) have shown similar benefits.

Delivering climate services using interoperable web services can lower the barriers for both data providers and data users. In particular, it can enhance the reusability of data and components in various applications, and get increased return on investment. For example, the WPS standard can help deliver services as simple as generating maps or graphs up to execution complex climate models on super computers. Another example is to help publishing legacy data. These data can be very useful to deliver time-series information but due to the fact that they are in formats that are outdated they are not used. By publishing these data with interoperable services, they are thus instantly available to all software, tools, and applications that have implemented such standards. It can then leverage the information power of these data. Finally, using standardized web services enable coupling models more easily (like coupling climate and hydrological models (Goodall et al., 2013)) and contributing to develop Earth System Models such as Earth System Modeling Framework (ESMF) (Collins et al., 2005; Hill et al., 2004), the GEO Model Web initiative (Nativi et al., 2013a,b,c), or Environmental Virtual Observatories (EVO) (Karpouzoglou et al., 2016).

From a scientific perspective, it can help increase the integrity of data underlying climate research. During the Climategate (i.e., illegally hacked emails from the University of East Anglia in November 2009 claiming that global warming was just a conspiracy), several scientific studies have reported the difficulty to reproduce research results because of missing or poor quality datasets (Brahic, 2010; Garud et al., 2014; Skrydstrup, 2013). Interoperable climate services together with corresponding technical and scientific capacities can play a crucial role in ensuring quality, integrity and availability of datasets and consequently promoting, contributing and supporting research activities and a trusted open science.

Finally, using interoperable services can answer challenges identified by several authors. It brings interactivity and facilitate climate data visualization (Nocke et al., 2008), it facilitates integration to determine the value of an ecosystem (Hungate and Hampton, 2012), explores how biodiversity evolves with a changing climate (Stefano Nativi et al., 2009), uses EO data for monitoring ECVs (Obregón et al., 2014), or publishes in-situ measurements using Sensor Web Enablement (SWE) suite of standards.

7.2. Challenges

Even if interoperability can lead to numerous promising benefits for spatially-enabling climate services, there are also a number of challenges to be addressed. Among them, we can mention: (1) data visualization; (2) data volume; (3) data dimensions (e.g., time and space); (4) capacity building; (5) institutional and policies (Bojinski et al., 2014; Härtwig and Müller, 2013; Nocke et al., 2008; Woolf et al., 2005).

Data visualization is an essential technology for presentation and communicating climate data. However, due the variety and complexity of climate data (e.g., mutli-dimensional, space and time) it is a challenging task. In particular, visualizing data quality and uncertainty is key as it can affect policy design (Otto et al., 2015). WMS-Q, QualityML and UncertML can be the basis for a standardized and interoperable visualization approach of uncertainty and quality of climate data.

Another challenge relates with the enormous volume of climate and the complexity of climate models that require efficient and powerful processing capacities. Clearly, climate data are part of the Big Data paradigm and new approaches for processing climate data are required. Distributed computing and code mobility are promising solutions to solve this issue (Nativi et al., 2015).

From an interoperability perspective, syntactical and structural are important steps towards full interoperability. However, effective data use requires semantic interoperability, which is still a major issue to tackle. Semantic interoperability enables systems to link external resources through standard reference mechanisms (see Linked Data), giving them the ability to exchange data with unambiguous shared meaning, enabling machine logic, inferring and knowledge generation (Hitzler and Janowicz, 2013; Kuhn et al., 2014; Schade et al., 2011).

Besides technical aspects, an important challenge relies on governance, policies, and institution. In term of governance, the GFCS is already in place and GEO and GFCS have identified mutual benefits and are currently making efforts to strengthen their cooperation. This emphasized by a recent joint activity of the GEO Work Programme that is aiming to improve the coordination between GEO and the GFCS to build linkages at the national and regional level between activities implemented under both frameworks and support the five GFCS priority areas⁶⁸. This can certainly help to reach commitment and get support to bring interoperability on the table. Open Data policies and Data Sharing Principles are spreading in various communities and this will strongly influence climate community as well as institutions that are providing data and delivering climate services.

Finally, an important challenge concerns the development of capacities at human, institutional, and infrastructure levels (Giuliani et al., 2013). Building capacities will help to reach large adoption, acceptation and commitment on data sharing principles. It will also strengthen the capacity of scientists to provide usable and understandable

⁶⁸ https://www.earthobservations.org/activity.php?id=95.

G. Giuliani et al.

information to decision makers and convince data holders to make available their data to a wide audience facilitating data discovery, access, and processing.

7.3. Perspectives and recommendations

Unlike the notion of Ecosystem Services (e.g., the benefits people obtain from ecosystems) the definition of Climate Services mostly concentrate on informative aspects to ensure that reliable climate science is effectively communicated to end-users. This terminology can be confusing because people also obtain benefits from climate, since it influences all aspects of human societies in a variety of ways. Climate data and information play an essential role for national development planning, managing risks and developing efficient mitigation and adaptation strategies. This requires that climate data and information can be integrated into various applications as well as policies.

Considering the importance of climate data, starting to enable an interoperable access to all ECVs appears to be crucial. All the data sources have been already identified by GCOS and they have been recognized as essential to support the work of IPCC and UNFCCC. Providing an interoperable access to these key data will certainly help the GFCS to gain further commitments and support. This can be done together with the Group on Earth Observations that is building GEOSS, dealing with Climate as a cross-cutting area of overarching importance giving discovery and access capabilities to various climate data⁶⁹. Moreover, with the brokering framework develop within GEOSS, it should be easy to engage the major ECV data providers and enable an interoperable access. Using the brokering approach will ease integration and facilitate multi-disciplinary approaches. This is even more important in the light of the ongoing global sustainability efforts such as the 2030 Global Agenda on Sustainable Development setting up Sustainable Development Goals (Brandi, 2015; Griggs et al., 2013; Lu et al., 2015); Planetary Boundaries (Rockstrom et al., 2009; Steffen et al., 2015); Nexus (de Strasser et al., 2016); or Natural Capital (Costanza et al., 1997; Guerry et al., 2015) approaches. In all these efforts, climate data and information are an essential component and consequently having interoperable climate services will greatly facilitate integration. The OGC has released a White paper⁷⁰ on "OGC Information Technology Standards for Sustainable Development" showing that OGC standards can be highly beneficial for enhancing data value chain related to these sustainability efforts. It is also a first step from data towards knowledge. Indeed, data and information are not sufficient to take good decisions (Ackoff, 1999; Rowley, 2007). Interoperability can help to build efficient data value chain, supporting most of the GFCS objectives previously mentioned. It can also facilitate providers and users to engage in delivering climate services and ultimately accelerate innovation (Brooks, 2013). In particular, it can help developing countries to access data and information that are not available, access computing and processing capabilities that they cannot afford (e.g., shared infrastructures, cloud computing), and facilitate the access to internet in low bandwidth conditions (e.g., dedicated data formats like Geopackage) (Rashidan and Musliman, 2015; Singh and Bermudez, 2013). Finally, interoperability can increase integrity, transparency, and foster collaboration. These are at the heart of the GFCS Ethical Framework for Climate Services⁷¹.

8. Conclusions

Addressing the challenges of climate change is critical for a more sustainable future. The Global Framework for Climate Services represents a global coordinated effort to improve the welfare of people vulnerable to climate variability and climate change.

Providing climate information that responds to users' needs and assists individuals and organizations in making decisions requires effective and efficient mechanisms. In particular, it should be coordinated in a coherent chain to create and build value. Climate data and information are essential to understand major environmental changes. It is crucial to make these data and information available not in the way that it is collected, but in the way that it is being used by the largest audience possible. Therefore, being able to easily combine climate data with other data sources is a major prerequisite to enable multi-disciplinary scientific analysis on the environment.

To deliver efficient climate services, interoperability is an essential element to consider extending the notion of service to provide a technical perspective as well. The GFCS should extend efforts towards more interoperable and web-based services. To answer this need, many open geospatial standards already exist and can be beneficial to strengthen the climate community to deliver interoperable climate services. These standards can be useful all along the data value chain (e.g., from data acquisition to decision-making) and can greatly enhance the scope of climate services. Enabling an interoperable access to all 50 Essential Climate Variables can help leveraging the full potential of climate services and can significantly contribute to the advancement of GFCS objectives and vision. Interoperable climate services have the potential to become the intelligence towards the transition to a climate-resilient society. They can support decision-makers taking informed decisions based on the best scientific knowledge to improve resilience, adaptation and mitigation strategies. Finally, thanks to interoperability, it can broaden the scope of climate services allowing to easily bring data into frameworks such as SDGs, Nexus, Natural Capitals, or Planetary Boundaries.

Acknowledgments

The authors would like to acknowledge the European Commission "Horizon 2020 Program" that funded the ERA-PLANET/GEOEssential project (Grant Agreement no. 689443).

The authors would like to acknowledge Veronica Grasso from the GFCS for her review of an earlier draft.

The views expressed in the paper are those of the authors and do not necessarily reflect the views of the institutions they belong to.

References

- Ackoff, R.L., 1999. From data to wisdom. In: J. W. & Sons (Ed.), Ackoff's Best. John Wiley & Sons, New York, pp. 170–172.
- Ariar Munoz, C., Oggioni, A., Brovelli, M. A., 2014. Geospatial web services for limnological data: a case study of sensor observation service for ecological observations (Vol. II-4, pp. 9–14). Presented at the ISPRS Annals of the Photogrammetry, Remote Sensing and Spatial Information Sciences.
- Bastin, L., Cornford, D., Jones, R., Heuvelink, G.B.M., Pebesma, E., Stasch, C., et al., 2013. Managing uncertainty in integrated environmental modelling: the UncertWeb framework. Environ. Model. Softw. 39, 116–134. http://dx.doi.org/10.1016/j.envsoft. 2012.02.008.
- Bates, J.J., Privette, J.L., 2012. A maturity model for assessing the completeness of climate data records. Eos, Trans. Am. Geophys. Union 93 (44). 441–441. https://doi. org/10.1029/2012EO440006.
- Baumann, P., Dehmel, A., Furtado, P., Ritsch, R., Widmann, N., 1999. Spatio-temporal retrieval with RasDaMan. Presented at the 25th VLDB Conference.
- Baumann, P., Mazzetti, P., Ungar, J., Barbera, R., Barboni, D., Beccati, A., et al., 2016. Big data analytics for earth sciences: the EarthServer approach. Int. J. Digital Earth 9 (1), 3–29. http://dx.doi.org/10.1080/17538947.2014.1003106.
- Béjar, R., Latre, M.A., Nogueras-Iso, J., Muro-Medrano, P.R., Zarazaga-Soria, F.J., 2009. Systems of systems as a conceptual framework for spatial data infrastructures. Int. J. Spatial Data Infrastruct. Res. 4, 17.
- Bernard, L., Ostlander, N., 2008. Assessing climate change vulnerability in the arctic using geographic information services in spatial data infrastructures. Clim. Change 87 (1–2), 263–281. http://dx.doi.org/10.1007/S10584-007-9346-0.

Bigagli, L., Nativi, S., 2013. NetCDF Uncertainty Conventions (NetCDF-U). Open Geospatial Consortium.

Blower, J.D., Alegre, R., Bennett, V.L., Clifford, D.J., Kershaw, P.J., Lawrence, B.N., 2013. Understanding climate data through commentary metadata: the charme project. In:

⁶⁹ http://www.earthobservations.org/climate.php.

⁷⁰ http://www.opengeospatial.org/pressroom/pressreleases/2188.

^{71 (}http://www.climate-services.org/wp-content/uploads/2015/09/CS-Ethics-White-Paper-Oct-2015.pdf).

Climate Services xxx (xxxx) xxx-xxx

Presented at the International Conference on Theory and Practice of Digital Libraries Springer, pp. 28–39.

Blower, J.D., Haines, K., Santokhee, A., Liu, C.L., 2009. GODIVA2: interactive visualization of environmental data on the Web. Philos. Trans. R. Soc. A 367 (1890), 1035–1039. http://dx.doi.org/10.1098/Rsta.2008.0180.

G. Giuliani et al.

- Blower, J.D., Maso, J., Diaz, D., Roberts, C.J., Griffiths, G.H., Lewis, J.P., et al., 2015. Communicating thematic data quality with web map services. Isprs Int. J. Geo-Inf. 4 (4), 1965–1981. http://dx.doi.org/10.3390/ijgi4041965.
- Bojinski, S., Verstraete, M., Peterson, T.C., Richter, C., Simmons, A., Zemp, M., 2014. The concept of essential climate variables in support of climate research, applications, and policy. Bull. Am. Meteorol. Soc. 95 (9), 1431–1443. http://dx.doi.org/10.1175/ BAMS-D-13-00047.1.
- Bosin, A., Dessi, N., Pes, B., 2011. Extending the SOA paradigm to e-Science environments. Fut. Generat. Comput. Syst. 27 (1), 20–31. http://dx.doi.org/10.1016/J. Future. 2010.07.003.
- Brahic, C., 2010. Climategate data sets to be made public. New Sci. 207 (2771), 5. http:// dx.doi.org/10.1016/S0262-4079(10)61822-2.
- Brandi, C., 2015. Safeguarding the earth system as a priority for sustainable development and global ethics: the need for an earth system SDG. J. Global Ethics 11 (1), 32–36. http://dx.doi.org/10.1080/17449626.2015.1006791.
- Brooks, M.S., 2013. Accelerating innovation in climate services: the 3 E's for climate service providers. Bull. Am. Meteorol. Soc. 94 (6), 807–819. http://dx.doi.org/10. 1175/Bams-D-12-00087.1.
- Buontempo, C., Hewitt, C.D., Doblas-Reyes, F.J., Dessai, S., 2014. Climate service development, delivery and use in Europe at monthly to inter-annual timescales. Clim. Risk Manage. 6, 1–5. http://dx.doi.org/10.1016/j.crm.2014.10.002.
- Buytaert, W., Baez, S., Bustamante, M., Dewulf, A., 2012. Web-based environmental simulation: bridging the gap between scientific modeling and decision-making. Environ. Sci. Technol. 46 (4), 1971–1976. http://dx.doi.org/10.1021/es2031278.
- Buyya, R., Yeo, C.S., Venugopal, S., Broberg, J., Brandic, I., 2009. Cloud computing and emerging IT platforms: vision, hype, and reality for delivering computing as the 5th utility. Fut. Generat. Comput. Syst. 25 (6), 599–616. http://dx.doi.org/10.1016/J. Future. 2008.12.001.
- Cannata, M., Antonovic, M., Molinari, M., Pozzoni, M., 2015. IstSOS, a new sensor observation management system: software architecture and a real-case application for flood protection. Geomat. Nat. Hazards Risk 6 (8), 635–650. http://dx.doi.org/10. 1080/19475705.2013.862572.
- Castronova, A.M., Goodall, J.L., Elag, M.M., 2013. Models as web services using the Open Geospatial Consortium (OGC) Web Processing Service (WPS) standard. Environ. Model. Softw. 41, 72–83. http://dx.doi.org/10.1016/j.envsoft.2012.11.010.
- Chengfang, H., Liping, D., & Wenli, Y., 2009. The research of interoperability in spatial catalogue service between CSW and THREDDS. Presented at the Geoinformatics, 2009 17th International Conference on. pp. 1–5. doi: http://dx.doi.org/10.1109/ geoinformatics.2009.5293503.
- Chinnachodteeranun, R., Honda, K., 2016. Sensor observation service API for providing gridded climate data to agricultural applications. Fut. Internet 8 (3), 40. http://dx. doi.org/10.3390/fi8030040.
- Christoph, M., Elke, S., van M. Jelle, G., Bart, S., Werner von, B., Arthur, H.W.B., 2016. Drivers and patterns of land biosphere carbon balance reversal. Environ. Res. Lett. 11 (4), 044002.
- CODATA, 2015. The Value of Open Data Sharing: A White Paper for the Group on Earth Observations, 42.
- Collaboration for Environmental Evidence, 2013. Guidelines for systematic review in environmental management. Centre for Evidence-Based Conservation, Bangor University, UK.
- Collins, N., Theurich, G., DeLuca, C., Suarez, M., Trayanov, A., Balaji, V., et al., 2005. Design and implementation of components in the Earth system modeling framework. Int. J. High Perform. Comput. Appl. 19 (3), 341–350. http://dx.doi.org/10.1177/ 1094342005056120.
- Costanza, R., D'Arge, R., De Groot, R., Farber, S., Grasso, M., Hannon, B., et al., 1997. The value of the world's ecosystem services and natural capital. Nature 387, 253–260.
- Cuca, B., 2016. Geospatial Future Is Open: Lessons Learnt from Applications Based on Open Data. In: Gervasi, O., Murgante, B., Misra, S., Rocha, C. A. M.A., Torre, C., Taniar, D. (Eds.), Computational Science and Its Applications - ICCSA 2016: 16th International Conference, Beijing, China, July 4-7, 2016, Proceedings, Part III. Springer International Publishing, Cham, pp. 491–502. Retrieved from doi: http://dx. doi.org/10.1007/978-3-319-42111-7_39.
- de Strasser, L., Lipponen, A., Howells, M., Stec, S., Bréthaut, C., 2016. A methodology to assess the water energy food ecosystems nexus in transboundary river basins. Water 8 (2), 59.
- Diamond, H.J., Lief, C.J., 2009. A comprehensive data portal for global climate information. Eos Trans. AGU 90 (39), 341–342. http://dx.doi.org/10.1029/ 2009EO390001.
- Diaz, P., Maso, J., Sevillano, E., Ninyerola, M., Zabala, A., Serral, I., Pons, X., 2012. Analysis of quality metadata in the GEOSS Clearinghouse. Int. J. Spatial Data Infrastruct. Res. 7, 352–377.
- Dolman, A.J., Belward, A., Briggs, S., Dowell, M., Eggleston, S., Hill, K., et al., 2016. A post-Paris look at climate observations. Nat. Geosci. 9 (9). 646–646. https://doi.org/ 10.1038/ngeo2785.
- Domenico, B., Caron, J., Davis, E., Kambic, R., Nativi, S., 2002. Thematic Real-time Environmental Distributed Data Services (THREDDS): Incorporating Interactive Analysis Tools into NSDL. J. Digit. Inf. 2 (4). Retrieved from https://journals.tdl.org/ jodi/index.php/jodi/article/view/51.

Domenico, B., Nativi, S., 2012. CF-netCDF3 Data Model Extension standard. Open Geospatial Consortium.

Domingos Freires Lucio, F., Grasso, V.F., 2016. The global framework for climate services

(GFCS). Clim. Serv. 2–3, 52–53. http://dx.doi.org/10.1016/j.cliser.2016.09.001. Dowell, M., Lecomte, P., Husband, R., Schulz, J., Mohr, T., Tahara, Y., 2013. Strategy

- Towards an Architecture for Climate Monitoring from Space. CEOS (p. 39). EUMETSAT, 2014. CORE-CLIMAX System Maturity Matrix Instruction Manual. p. 35. Darmstadt, Germany.
- European Commission, 2007. Directive 2007/2/EC of the European Parliament and the Council of 14 March 2007 establishing an Infrastructure for Spatial Information in the European Community (INSPIRE), 14.
- European Commission, 2015. A European research and innovation Roadmap for Climate Services.
- Garud, R., Gehman, J., Karunakaran, A., 2014. Boundaries, breaches, and bridges: the case of Climategate. Res. Policy 43 (1), 60–73. http://dx.doi.org/10.1016/j.respol. 2013.07.007.
- Geraci, A., 1991. IEEE standard computer dictionary: a compilation of IEEE standard computer glossaries. IEEE Std. 610, 1–217. http://dx.doi.org/10.1109/IEEESTD. 1991.106963.
- Gijzen, H., 2013. Big data for a sustainable future. Nature 502 (7469), 38.
- Giorgi, F., Jones, C., Asrar, G.R., 2009. Addressing climate information needs at the regional level: the CORDEX framework. World Meteorol. Org. Bull. 58 (3), 175.
- Giuliani, G., Guigoz, Y., Lacroix, P., Ray, N., Lehmann, A., 2016. Facilitating the production of ISO-compliant metadata of geospatial datasets. Int. J. Appl. Earth Obs. Geoinf. 44, 239–243. http://dx.doi.org/10.1016/j.jag.2015.08.010.
- Giuliani, G., Lacroix, P., Guigoz, Y., Roncella, R., Bigagli, L., Santoro, M., et al., 2016. Bringing GEOSS services into practice: a capacity building resource on spatial data infrastructures (SDI). Trans. GIS. http://dx.doi.org/10.1111/tgis.12209. n/a-n/a.
- Giuliani, G., Nativi, S., Lehmann, A., Ray, N., 2012. WPS mediation: an approach to process geospatial data on different computing backends. Comput. Geosci. 47, 20–33. http://dx.doi.org/10.1016/j.cageo.2011.10.009.
- Giuliani, G., Ray, N., & Lehmann, A., 2013. Building Regional Capacities for GEOSS and INSPIRE: a Journey in the Black SeaCatchment. International Journal of Advanced Computer Science and Applications, EnviroGRIDS Special Issue on "Building a Regional Observation System in the Black Sea Catchment", 19–27. https://doi.org/ 10.14569/SpecialIssue.2013.030302.
- Goodall, J.L., Saint, K.D., Ercan, M.B., Briley, L.J., Murphy, S., You, H., et al., 2013. Coupling climate and hydrological models: interoperability through Web Services. Environ. Model. Softw. 46, 250–259. http://dx.doi.org/10.1016/j.envsoft.2013.03. 019.
- Gray, J., Bosworth, A., Layman, A., Pirahesh, H., 1995. Data Cube: A Relational Aggregation Operator Generalizing Group-By, Cross-Tab, and Sub-Totals. pp. 152–159.
- Griggs, D., Stafford-Smith, M., Gaffney, O., Rockstrom, J., Ohman, M.C., Shyamsundar, P., Noble, I., 2013. Sustainable development goals for people and planet. Nature 495 (7441), 305–307.
- Grothe, M., Brentjens, T., 2013. Joining Tabular and Geographic Data Merits and Possibilities of the Table Joining Service. Geonovum, Amersfoort, The Netherlands, pp. 53.
- Guerry, A.D., Polasky, S., Lubchenco, J., Chaplin-Kramer, R., Daily, G.C., Griffin, R., et al., 2015. Natural capital and ecosystem services informing decisions: from promise to practice. Proc. Natl. Acad. Sci. U.S.A. 112 (24), 7348–7355. http://dx.doi.org/10. 1073/pnas.1503751112.
- Harold, J., Lorenzoni, I., Shipley, T.F., Coventry, K.R., 2016. Cognitive and psychological science insights to improve climate change data visualization. Nat. Clim. Change 6 (12), 1080–1089. http://dx.doi.org/10.1038/nclimate3162.
- Härtwig, M., Müller, M., 2013. SDI components for ad-hoc analysis of climatological timeseries data. Presented at the 16th AGILE International Conference on Geographic Information Science.
- Hewitt, C., Mason, S., Walland, D., 2012. COMMENTARY: the global framework for climate services. Nat. Clim. Change 2 (12), 831–832.
- Hill, C., DeLuca, C., Balaji, Suarez, M., da Silva, A., 2004. The architecture of the earth system modeling framework. Comput. Sci. Eng. 6 (1), 18–28. http://dx.doi.org/10. 1109/Mcise.2004.1255817.
- Hitzler, P., Janowicz, K., 2013. Linked data, big data, and the 4th paradigm. Semantic Web 4 (3), 233–235. http://dx.doi.org/10.3233/SW-130117.
- Hoff, H., 2011. Understanding the nexus: background paper for the Bonn 2011 Nexus Conference.
- Hollmann, R., Merchant, C.J., Saunders, R., Downy, C., Buchwitz, M., Cazenave, A., et al., 2013. The Esa climate change initiative satellite data records for essential climate variables. Bull. Am. Meteorol. Soc. 94 (10), 1541–1552. http://dx.doi.org/10.1175/ BAMS-D-11-00254.1.
- Horsburgh, J.S., Jones, A.S., Stevens, D.K., Tarboton, D.G., Mesner, N.O., 2010. A sensor network for high frequency estimation of water quality constituent fluxes using surrogates. Environ. Model. Softw. 25 (9), 1031–1044. http://dx.doi.org/10.1016/J. Envsoft. 2009.10.012.
- Huges, G., 2011. The Global Framework for Climate Services. Retrieved from. http:// www.earthzine.org/2011/05/12/the-global-framework-for-climate-services/.
- Hungate, B.A., Hampton, H.M., 2012. Ecosystem services valuing ecosystems for climate. Nat. Clim. Change 2 (3), 151–152.
- Inman, M., 2011. Opening the future. Nat. Clim. Change 1 (1), 7–9. http://dx.doi.org/10. 1038/nclimate1058.
- Karpouzoglou, T., Zulkafli, Z., Grainger, S., Dewulf, A., Buytaert, W., Hannah, D.M., 2016. Environmental Virtual Observatories (EVOS): prospects for knowledge co-creation and resilience in the Information Age. Curr. Opin. Environ. Sust. 18, 40–48. http:// dx.doi.org/10.1016/j.cosust.2015.07.015.
- Kaye, N.R., Hartley, A., Hemming, D., 2012. Mapping the climate: guidance on appropriate techniques to map climate variables and their uncertainty. Geosci. Model Dev. 5 (1), 245–256. http://dx.doi.org/10.5194/Gmd-5-245-2012.

G. Giuliani et al.

Climate Services xxx (xxxx) xxx-xxx

- Klein, T., Langner, J., Frankenberg, B., Svensson, J., Broman, J., Bennet, C., Langborg, T., 2013. ECDS - A swedish research infrastructure for the open sharing of environment and climate data. Data Sci. J. 12.
- Kotova, L., Manez Costa, M., Rodriguez Perez, M.J., Whiffin, F., Garrett, N., Bessembinder, J., et al., 2017. The first Climateurope Festival: climate information at your service. Clim. Serv. http://dx.doi.org/10.1016/j.cliser.2017.07.005.
- Kristin, P., Andrew, W., Peer-Timo, B., Dean, W., Charles, D., Valerio, P., Chris, J., 2009. Visualization of uncertainty and ensemble data: exploration of climate modeling and weather forecast data with integrated ViSUS-CDAT systems. J. Phys: Conf. Ser. 180 (1), 012089.
- Kuhn, W., Kauppinen, T., Janowicz, K., 2014. Linked Data a paradigm shift for geographic information science. In: Duckham, M., Pebesma, E., Stewart, K., Frank, A. (Eds.), Geographic Information Science. 8728. Springer International Publishing, pp. 173–186. Retrieved from http://dx.doi.org/10.1007/978-3-319-11593-1_12 http:// link.springer.com/chapter/10.1007%2F978-3-319-11593-1_12.
- Kutschera, P., Bartha, M., Havlik, D., 2011. SUDPLAN's experiences with the OGC-based model web services for the climate change usage area. In: Environmental Software Systems. Frameworks of eEnvironment Springer, Berlin, Heidelberg, pp. 589–604.
- Laney, C.M., Pennington, D.D., Tweedie, C.E., 2015. Filling the gaps: sensor network use and data-sharing practices in ecological research. Front. Ecol. Environ. 13 (7), 363–368. http://dx.doi.org/10.1890/140341.
- Laniak, G.F., Olchin, G., Goodall, J., Voinov, A., Hill, M., Glynn, P., et al., 2013. Integrated environmental modeling: a vision and roadmap for the future. Environ. Model. Softw. 39, 3–23. http://dx.doi.org/10.1016/j.envsoft.2012.09.006.
- Lee, J.G., Kang, M., 2015. Geospatial big data: challenges and opportunities. Big Data Res. 2 (2), 74–81. http://dx.doi.org/10.1016/j.bdr.2015.01.003.
- Lehmann, A., Chaplin-Kramer, R., Lacayo, M., Giuliani, G., Thau, D., Koy, K., et al., 2017. Lifting the information barriers to address sustainability challenges with data from physical geography and earth observation. Sustainability 9 (5), 858. http://dx.doi. org/10.3390/su9050858.
- Lehmann, A., Giuliani, G., Mancuso, E., Abbaspour, K., Sözen, S., Gorgan, D., et al., 2014. Filling the gap between Earth observation and policy making in the Black Sea catchment with enviroGRIDS. Environ. Sci. Policy.
- Lehmann, A., Giuliani, G., Ray, N., Rahman, K., Abbaspour, K.C., Nativi, S., Beniston, M., 2014. Reviewing innovative Earth observation solutions for filling science-policy gaps in hydrology. J. Hydrol. 518 (Part B(0)), 267–277. http://dx.doi.org/10.1016/j. ihydrol.2014.05.059.
- Lourenço, T.C., Swart, R., Goosen, H., Street, R., 2016. The rise of demand-driven climate services. Nat. Clim. Change 6 (1), 13–14. http://dx.doi.org/10.1038/nclimate2836.
- Lu, Y.L., Nakicenovic, N., Visbeck, M., Stevance, A.S., 2015. Five priorities for the UN sustainable development goals. Nature 520 (7548), 432–433.
- Mazzetti, P., Roncella, R., Mihon, D., Bacu, V., Lacroix, P., Guigoz, Y., et al., 2016. Integration of data and computing infrastructures for earth science: an image mosaicking use-case. Earth Sci. Inf. 1–18. http://dx.doi.org/10.1007/s12145-016-0255-5.
- Medri, S., Banos de Guisasola, E., Gualdi, S., 2012. Overview of the Main International Climate Services (SSRN Scholarly Paper No. ID 2194841). Social Science Research Network, Rochester, NY. Retrieved from https://papers.ssrn.com/abstract = 2194841.
- Miller, H.G., Mork, P., 2013. From data to decisions: a value chain for big data. It Profess. 15 (1), 57–59.
- Moges, H.-T., Vlasselaer, V.V., Lemahieu, W., Baesens, B., 2016. Determining the use of data quality metadata (DQM) for decision making purposes and its impact on decision outcomes – an exploratory study. Decis. Support Syst. 83, 32–46. http://dx.doi. org/10.1016/j.dss.2015.12.006.
- National Geospatial Advisory Committee Landsat Advisory Group, 2014. The Value Proposition for Landsat Applications 2014 Update, 11.
- National Research Council, Mapping Science Committee, 1993. Toward a coordinated spatial data infrastructure for the nation. National Academies Press.
- Nativi, S., Bigagli, L., 2009. Discovery, mediation, and access services for earth observation data. IEEE J. Selected Top. Appl. Earth Observ. Remote Sens. 2 (4), 233–240. http://dx.doi.org/10.1109/jstars.2009.2028584.
- Nativi, S., Craglia, M., Pearlman, J., 2012. The brokering approach for multidisciplinary interoperability: a position paper. Int. J. Spatial Data Infrastruct. Res. 7, 1–15.
- Nativi, S., Craglia, M., Pearlman, J., 2013. Earth science infrastructures interoperability: the brokering approach. IEEE J. Selected Top. Appl. Earth Observ. Remote Sens. 6 (3), 1118–1129. http://dx.doi.org/10.1109/jstars.2013.2243113.
- Nativi, S., Domenico, B., 2009. Enabling interoperability for Digital Earth: Earth Science coverage access services. Int. J. Digital Earth 2, 79–104. http://dx.doi.org/10.1080/ 17538940902866179.
- Nativi, S., Domenico, B., Caron, J., Davis, E., Bigagli, L., 2006. Extending THREDDS middleware to serve OGC community. Adv. Geosci. 8, 57–62. http://dx.doi.org/10. 5194/adgeo-8-57-2006.
- Nativi, S., Mazzetti, P., Craglia, M., Pirrone, N., 2013. The GEOSS solution for enabling data interoperability and integrative research. Environ. Sci. Pollut. Res.
- Nativi, S., Mazzetti, P., Geller, G.N., 2013. Environmental model access and interoperability: the GEO Model Web initiative. Environmental Modelling and Software 39, 214–228. http://dx.doi.org/10.1016/j.envsoft.2012.03.007.
- Nativi, S., Mazzetti, P., Saarenmaa, H., Kerr, J., et al., 2009. Biodiversity and climate change use scenarios framework for the GEOSS interoperability pilot process. Ecol. Inform. 4 (1), 23–33. http://dx.doi.org/10.1016/j.ecoinf.2008.11.002.
- Nativi, S., Mazzetti, P., Santoro, M., Papeschi, F., Craglia, M., Ochiai, O., 2015. Big Data challenges in building the global earth observation system of systems. Environ. Model. Softw. 68, 1–26. http://dx.doi.org/10.1016/j.envsoft.2015.01.017.
- Nocke, T., Sterzel, T., Böttinger, M., Wrobel, M., 2008. Visualization of Climate and Climate Change Data: An Overview. Presented at the Digital Earth Summit on

Geoinformatics 2008: Tools for Global Change Research, pp. 226-232.

- Obregón, A., Nitsche, H., Körber, M., Kreis, A., Bissolli, P., Friedrich, K., Rösner, S., 2014. Satellite-based climate information within the WMO RA VI Regional Climate Centre on Climate Monitoring. Adv. Sci. Res. 11, 25–33. http://dx.doi.org/10.5194/asr-11-25-2014.
- Open Geospatial Consortium, 2004. The Havoc of Non-Interoperability. 7.
- Open Geospatial Consortium, 2006a. OpenGIS Web Map Server Implementation Specification. Retrieved from http://www.opengeospatial.org/standards/wms, p. 85.
- Open Geospatial Consortium, 2006b. Web Coverage Service (WCS) Implementation Specification. 143. Retrieved from http://www.opengeospatial.org/standards/wcs.
- Open Geospatial Consortium, 2007a. OpenGIS Catalogue Services Specification. Retrieved from http://www.opengeospatial.org/standards/cat. p. 218.
- Open Geospatial Consortium, 2007b. OpenGIS Web Processing Service. Retrieved from http://www.opengeospatial.org/standards/wps. p. 87.
- Open Geospatial Consortium, 2010a. OpenGIS Georeferences Table Joining Service (TJS) Implementation Standards. OGC p. 92.
- Open Geospatial Consortium, 2010b. OpenGIS Web Feature Service 2.0 Interface Standard.
- Ostensen, O., O'Brien, D., Cooper, A., 2008. Measurements to know and understand our world. ISO Focus, (February), 35–37.
- Otto, F.E.L., Frame, D.J., Otto, A., Allen, M.R., 2015. Embracing uncertainty in climate change policy. Nat. Clim. Change 5 (10), 917. http://dx.doi.org/10.1038/ NCLIMATE2716.
- Overpeck, J.T., Meehl, G.A., Bony, S., Easterling, D.R., 2011. Climate data challenges in the 21st century. Science 331 (6018), 700–702. http://dx.doi.org/10.1126/Science. 1197869.

Palmer, T.N., 2000. Predicting uncertainty in forecasts of weather and climate. Rep. Prog. Phys. 63 (2), 71.

- Pebesma, E., Cornford, D., Dubois, G., Heuvelink, G.B.M., Hristopulos, D., Pilz, J., et al., 2011. INTAMAP: the design and implementation of an interoperable automated interpolation web service. Comput. Geosci. 37 (3), 343–352. http://dx.doi.org/10. 1016/J.Cageo. 2010.03.019.
- Plummer, R., de Loë, R., Armitage, D., 2012. A systematic review of water vulnerability assessment tools. Water Resour. Manage 26 (15), 4327–4346. http://dx.doi.org/10. 1007/s11269-012-0147-5.
- Preuschmann, S., Hänsler, A., Kotova, L., Dürk, N., Eibner, W., Waidhofer, C., et al., n.d. The IMPACT2C web-atlas – Conception, organization and aim of a web-based climate service product. Clim. Serv. doi: http://dx.doi.org/10.1016/i.cliser.2017.03.005.
- Pullin, A.S., Stewart, G.B., 2006. Guidelines for systematic review in conservation and environmental management. Conserv. Biol. 20 (6), 1647–1656. http://dx.doi.org/10. 1111/i.1523-1739.2006.00485.x.
- Rashidan, M.H., Musliman, I.A., 2015. GeoPackage as future ubiquitous GIS data format: a review. Jurnal Teknologi 73 (5).
- Rockstrom, J., Steffen, W., Noone, K., Persson, A., Chapin, F.S., Lambin, E., 2009. Planetary boundaries: exploring the safe operating space for humanity. Ecol. Soc. 14 (2) Retrieved from://WOS:000278707200010.
- Rodila, D., Gorgan, D., 2012. Geospatial and grid interoperability through OGC services gridification. IEEE J. Selected Top. Appl. Earth Observ. Remote Sens. 5 (6), 1650–1658. http://dx.doi.org/10.1109/Jstars.2012.2217115.
- Rodila, D., Ray, N., Gorgan, D., 2015. Conceptual model for environmental science applications on parallel and distributed infrastructures. Environ. Syst. Res. 4 (1), 1–16. http://dx.doi.org/10.1186/s40068-015-0050-1.
- Rowley, J., 2007. The wisdom hierarchy: representations of the DIKW hierarchy. J. Inform. Sci. 33 (2), 163–180. http://dx.doi.org/10.1177/0165551506070706. Ryan, B., 2016. The benefits from open data are immense. Geospatial World 72–73.
- Scanlon, T., Nightingale, J., Muller, J.-P., Boersma, F., De Rudder, A., Lambert, J., 2015. QA4ECV: A robust quality assurance service for terrestrial and atmospheric ECVs and ECV precursors. Presented at the RSPSoc, NCEO and CEOI-ST Joint Annual Conference.
- Schade, S., Granell, C., Diaz, L., 2011. Augmenting SDI with Linked Data.
- Sessa, R., Latham, J., 2008. Systematic observations for assessing, mitigating and adapting to climate change. ISO Focus, (February), 30–34.
- Shujia, Z., Cruz, C., Duffy, D., Tucker, R., & Purcell, M., 2010. Accelerating Climate and Weather Simulations Through Hybrid Computing. In M. Parashar, R. Buyya, C. D. D. E. of N. G. C. Shujia Zhou; Cruz, G. M. D. U. S. A. Nasa Goddard Space Flight Center, & R. P. M. I. B. M. I. L. I. Tucker (Eds.). Presented at the 2010 10th IEEE/ACM International Conference on Cluster, Cloud and Grid Computing (CCGrid), IEEE Computer Society, pp. 797–801. Doi: http://dx.doi.org/10.1109/ccgrid.2010.75.
- Singh, R., Bermudez, L., 2013. Emerging Geospatial Sharing Technologies in Earth and Space Science Informatics (Vol. 1, p. 08). Presented at the AGU Fall Meeting Abstracts.
- Skrydstrup, M., 2013. Tricked or troubled natures? How to make sense of "climategate". Environ. Sci. Policy 28, 92–99. http://dx.doi.org/10.1016/j.envsci.2012.11.012.
- Srinivasan, G., Boscolo, R., Cegnar, T., Coelho, C., Fuchs, T., Kamga, A., et al., 2015. Climate services – transitioning from CLIPS to GFCS. WMO Bull. 64 (1), 23–27.
- Stasch, C., Foerster, T., Autermann, C., Pebesma, E., 2012. Spatio-temporal aggregation of European air quality observations in the Sensor Web. Comput. Geosci. 47, 111–118. http://dx.doi.org/10.1016/j.cageo.2011.11.008.
- Steffen, W., Richardson, K., Rockström, J., Cornell, S.E., Fetzer, I., Bennett, E.M., et al., 2015. Planetary boundaries: guiding human development on a changing planet. Science. http://dx.doi.org/10.1126/science.1259855.
- Stephens, A., James, P., Alderson, D., Pascoe, S., Abele, S., Iwi, A., Chiu, P., 2012. The challenges of developing an open source, standards-based technology stack to deliver the latest UK climate projections. Int. J. Digital Earth 5 (1), 43–62. http://dx.doi.org/ 10.1080/17538947.2011.571724.
- Street, R.B., 2015. Towards a leading role on climate services in Europe: a research and

G. Giuliani et al.

- innovation roadmap. Clim. Serv. http://dx.doi.org/10.1016/j.cliser.2015.12.001. Strobl, J., Belgiu, M., Nazarkulova, A., 2011. Building an SDI as a community project – challenges in emerging economies. Int. J. Spatial Data Infrastruct. Res.
- Sun, X., Shen, S., Leptoukh, G.G., Wang, P., Di, L., Lu, M., 2012. Development of a Webbased visualization platform for climate research using Google Earth. Comput. Geosci. 47, 160–168. http://dx.doi.org/10.1016/j.cageo.2011.09.010.
- Swart, R.J., de Bruin, K., Dhenain, S., Dubois, G., Groot, A., von der Forst, E., 2017. Developing climate information portals with users: promises and pitfalls. Clim. Serv. http://dx.doi.org/10.1016/j.cliser.2017.06.008.
- Trenberth, K.E., Marquis, M., Zebiak, S., 2016. The vital need for a climate information system. Nat. Clim. Change 6 (12), 1057–1059. http://dx.doi.org/10.1038/ nclimate3170.
- UN Global Pulse, 2012. Big Data for Development: Challenges & Opportunities. p. 47. New York.
- UNFCCC, 2015. Adoption of the Paris agreement.
- United Nations, 2012. The future we want, Nations, United 53.
- United Nations, 2015. Transforming our world: the 2030 Agenda for Sustainable Development, A/RES/70/1 Nations, United 35.
- Valcke, S., Craig, A., Dunlap, R., Riley, G.D., 2015. Sharing experiences and outlook on coupling technologies for earth system models. Bull. Am. Meteorol. Soc. 97 (3), ES53–ES56. http://dx.doi.org/10.1175/BAMS-D-15-00239.1.
- Vaughan, C., Buja, L., Kruczkiewicz, A., Goddard, L., 2016. Identifying research priorities to advance climate services. Clim. Serv. 4, 65–74. http://dx.doi.org/10.1016/j.cliser. 2016.11.004.
- Vaughan, C., Dessai, S., 2014. Climate services for society: origins, institutional arrangements, and design elements for an evaluation framework. Wiley Interdiscip. Rev.: Clim. Change 5 (5), 587–603. http://dx.doi.org/10.1002/wcc.290.

- Wagemann, J., Clements, O., Figuera, R.M., Rossi, A.P., Mantovani, S., 2017. Geospatial web services pave new ways for server-based on-demand access and processing of Big Earth Data. Int. J. Digital Earth 1–19. http://dx.doi.org/10.1080/17538947.2017. 1351583.
- Wessels, B., Finn, R.L., Linde, P., Mazzetti, P., Nativi, S., Riley, S., et al., 2014. Issues in the development of open access to research data. Prometheus 32 (1), 49–66. http:// dx.doi.org/10.1080/08109028.2014.956505.
- Wilhelmi, O.J., Boehnert, J., Sampson, K., 2016. Visualizing the climate's future. EOS 97. http://dx.doi.org/10.1029/2016EO042207.
- Woolf, A., Cramer, R., Gutierrez, M., van Dam, K.K., Kondapalli, S., Latham, S., et al., 2005. Standards-based data interoperability in the climate sciences. Meteorol. Appl. 12 (1), 9–22. http://dx.doi.org/10.1017/S1350482705001556.
- Woolf, A., Lawrence, B., Lowry, R., Kleese van Dam, K., Cramer, R., Gutierrez, M., et al., 2006. Data integration with the Climate Science Modelling Language. Adv. Geosci. 8, 83–90. http://dx.doi.org/10.5194/adgeo-8-83-2006.
- World Meteorlogical Organization, 2011. Climate Knowledge for Action: A global framework for climate services – empowering the most vulnerable.
- World Meteorlogical Organization, 2012. Climate Exchange. p. 290.
- World Meteorlogical Organization, 2015. Seventeenth World Meteorological Congress Abridged final report with resolutions. p. 708.
- World Meteorlogical Organization, 2016. WMO Statement on the Status of the Global Climate in 2015. WMO, Geneva, Switzerland, pp. 28.
- Zabala, A., Riverola, A., Serral, I., Diaz, P., Lush, V., Maso, J., et al., 2013. Rubric-Q: adding quality-related elements to the GEOSS clearinghouse datasets. IEEE J. Selected Top. Appl. Earth Observ. Remote Sensing 6 (3), 1676–1687. http://dx.doi. org/10.1109/Jstars.2013.2259580.

Climate Services xxx (xxxx) xxx-xxx